TANDBERG DATA Securing your Information



TANDBERG DATA TANDBERG DATA

Product Manual

RDX[®] QuikStation[™]

Part Number 1019786 Rev. B October 2011

www.tandbergdata.com

| Copyright | Copyright contained reproduce language o magnetic, permission | ht 2011 by Tandberg Data. All rights reserved. This item and the information d herein are the property of Tandberg Data. No part of this document may be ced, transmitted, transcribed, stored in a retrieval system, or translated into any e or computer language in any form or by any means, electronic, mechanical, c, optical, chemical, manual, or otherwise, without the express written on of Tandberg Data. | | |
|-------------------------|--|--|---------------------|--|
| DISCLAIMER | Tandberg Data makes no representation or warranties with respect to the contents of this document and specifically disclaims any implied warranties of merchantability or fitness for any particular purpose. Further, Tandberg Data reserves the right to revise this publication without obligation of Tandberg Data to notify any person or organization of such revision or changes. | | | |
| Trademark Notices | Tandberg Data StorageLibrary, StorageLoader, SecureService, DPS1000 Series, DPS 2000, VXA, SLR, RDX QuikStor, RDX QuikStation, AccuVault, and AccuGuard are trademarks of Tandberg Data. RDX is a registered trademark of Tandberg Data S.a.r.l. All other product names are trademarks or registered trademarks of their respective owners. | | | |
| Part Number | 1019786 Re | evision B | | |
| REVISION HISTORY | Revision | Date | Description | |
| | А | May 2011 | Initial release. | |
| | В | October 2011 | Version 2.0 updates | |
| | | | | |

Note: The most current information about this product is available at Tandberg Data's web site (www.tandbergdata.com).

П

1019786

| Product Warranty Caution | The RDX QuikStation by Tandberg Data Corporation is warranted to be free from defects in materials, parts, and workmanship and will conform to the current product specification upon delivery. For the specific details of your warranty, refer to your sales contract or see the Tandberg Data web site (www.tandbergdata.com). | | |
|--------------------------------|---|--|--|
| | The warranty for the appliance shall not apply to failures caused by: | | |
| | Physical abuse or use not consistent with the operating instructions or product specifications. | | |
| | Repair or modification by any one other than Tandberg Data's personnel or agent in a manner differing from the maintenance instructions provided by Tandberg Data. | | |
| | Removal of the Tandberg Data identification label(s). | | |
| | Physical abuse due to improper packaging of returned unit. | | |
| | If problems with the RDX QuikStation occur, contact your maintenance organization; do not void the product warranty by allowing untrained or unauthorized personnel to attempt repairs. | | |
| A Caution | Returning the RDX QuikStation in unauthorized packaging may damage the unit and void the warranty. If you are returning the unit for repair, package it in its original packaging (or in replacement packaging obtained from your vendor.) | | |
| Contacting Tandberg Data | Visit the Support section of the Tandberg Data web site (www.tandbergdata.com) for information on contacting technical support. | | |

TANDBERG DATA CORPORATION

RDX QUIKSTATION

NOTES

CONTENTS

| A | bout This Manual ix |
|---|---|
| 1 | Product Overview 1 |
| | RDX QuikStation Features2 |
| | Configuration Flexibility 2 |
| | Tandberg Data RDX Removable Disk Technology 2 |
| | High Performance and Capacity 3 |
| | Compatible with Major Backup Software |
| | Integrated Web-based Management |
| | Dual Gigabit Ethernet |
| | Compact Design 3 |
| | RDX QuikStation Components |
| | RDX Dock LCD Panels 6 |
| | RDX Cartridge-Eject button/Power LED |
| | RDX QuikStation LED Indicator Panel |
| 2 | Installation and Setup 9 |
| | Preparing for Installation |
| | Verifying Accessory Kit Contents |
| | Installing the RDX QuikStation in a Rack12 |
| | Assembling the Outer Rails |
| | Installing the Outer Rails into the Rack |
| | Attaching the Inner Rails |
| | Installing the RDX QuikStation into the Rack |
| | Connecting Network and Power Cables |
| | Configuring Network Information |
| | Connecting to the Remote Management Console |

| | Configuring the RDX QuikStation | 21 |
|---|---|----|
| | Logical Device Type | 21 |
| | System Settings | 22 |
| 3 | iSCSI Configuration | 23 |
| | iSCSI Configuration Overview | 24 |
| | Connecting to iSCSI Targets Without Authentication | 25 |
| | Connecting iSCSI Targets for Windows | 25 |
| | Connecting iSCSI Targets for Linux | 28 |
| | Connecting iSCSI Targets for Macintosh | 30 |
| | Setting iSCSI Security | 33 |
| | Setting iSCSI Security in the Remote Management Console | 34 |
| | Connecting iSCSI Targets with CHAP for Windows | 37 |
| | Connecting iSCSI Targets with CHAP for Linux | 40 |
| | Connecting iSCSI Targets with CHAP for Macintosh | 42 |
| 4 | System Configuration | 47 |
| | System Settings Overview. | 48 |
| | Setting Date and Time | 49 |
| | Configuring Network Information | 50 |
| | Configuring Advanced Network Settings | 51 |
| | Setting Email Notifications | 53 |
| | Setting Basic Email notification | 53 |
| | Setting SMTP Authentication | 53 |
| | Customizing Email Subject and "From" Address | 54 |
| | Setting RDX QuikStation Options | 54 |
| | Setting Removable Disk Options | 56 |
| | Setting the Auto-Insert Option | 57 |
| | Managing User Groups | 60 |
| | Group Permissions Overview | 61 |
| | Adding Users to a Group | 61 |
| | Removing Users from a Group | 62 |
| | Editing User Information | 63 |
| | Recovering the Built-In Administrator Password | 64 |
| | Resetting the Password via Email | 64 |
| | Resetting the Password via USB | 66 |

| 5 | Logical Device Management | 69 |
|---|---|----|
| | Logical Device Type Overview | 70 |
| | Choosing a Logical Device Type | 71 |
| | iSCSI Device Overview | 72 |
| | Customizing the iSCSI Devices View | 73 |
| | Managing Disk and Tape Drive Configurations | 74 |
| | Viewing Logical Properties for RDX or Virtual Tape Drives | 74 |
| | Viewing Active iSCSI Connections for RDX or Tape Drives | 75 |
| | Managing Tape Library and Autoloader Configurations | 76 |
| | Viewing Logical Properties for Tape Libraries/Autoloaders | 76 |
| | Viewing Active iSCSI Connections for Tape Libraries/Autoloaders 77 | |
| | Moving a Media Cartridge | 78 |
| | Setting the Unique Inquiry Option | 80 |
| | Setting the Autoloader Option | 80 |
| 6 | Physical Device Management | 83 |
| | Viewing Physical Device Properties | 84 |
| | Ejecting RDX Cartridges | 86 |
| | Erasing RDX Cartridges | 86 |
| | Cloning Data | 87 |
| 7 | System Maintenance | 91 |
| | System Maintenance Overview | 92 |
| | Performing an LCD Diagnostic Test | 93 |
| | LCD Diagnostic Test Sequence | 94 |
| | Performing a Dock Diagnostic Test | 95 |
| | Generating System Logs | 96 |
| | Restarting and Shutting Down the RDX QuikStation | 97 |
| | Exporting System Configuration Files | 98 |
| | Importing RDX QuikStation System Configuration Files | 98 |
| | Manually Updating Firmware | 99 |
| | Using a USB Flash Drive for System Tasks | 01 |
| A | Troubleshooting 10 | 03 |
| | RDX QuikStation Dock Error Codes | 03 |
| | Basic Troubleshooting Information1 | 04 |

| B Specifications |
|---|
| Physical Specifications |
| Power Specifications |
| Environmental Specifications |
| Safety and Regulatory Agency Compliance |
| Underwriters Laboratory 107 |
| GS 107 |
| European Community 107 |
| United States: FCC Declaration of Conformity |
| Canadian Verification |
| Japan |
| Australia and New Zealand |
| Waste of Electronic and Electrical Equipment (WEEE) 109 |
| China RoHS 109 |
| C Safety Guidelines 111 |
| General Safety Precautions |
| Electrical Safety Precautions |
| ESD Precautions |
| Operating Precautions |
| Index |

ABOUT THIS MANUAL

This manual describes how to install and operate the RDX QuikStation and is intended for use by anyone deploying the RDX QuikStation into their network environment. Familiarity with system and network configuration is highly recommended.

The following chapters are included in this manual:

Overview

 Chapter 1 provides an overview of the RDX QuikStation's features and components.

Installation setup

- Chapter 2 describes information on how to rackmount, connect, and initially configure the RDX QuikStation.
- Chapter 3 describes how to connect host computers to RDX QuikStation iSCSI targets, including how to set up iSCSI security.

Using the RDX Remote Management Console

- Chapter 4 describes RDX QuikStation system tasks, such as network configuration, and user account management.
- Chapter 5 provides information about the logical device interface in the Remote Management Console.
- Chapter 6 provides information about the physical device interface in the Remote Management Console.
- Chapter 7 describes RDX QuikStation system maintenance tasks, such as performing diagnostic tests, cloning data, and updating firmware.

Supplemental Information

- Appendix A provides error code definitions and some basic troubleshooting information.
- Appendix B provides product specifications.
- Appendix C describes important safety precautions to take when using the RDX QuikStation.

RELATED PUBLICATIONS

For additional information about the RDX QuikStation, refer to the following publications, which are available at www.tandbergdata.com.

- RDX QuikStation Quick Start Guide, part number 1019372
- Field Instruction: Installing the RDX QuikStation in a Rack, part number 1019782
- RDX QuikStation Knowledge Base articles available online at www.tandbergdata.com

CONVENTIONS USED IN THIS MANUAL

This manual uses the following conventions:

Note: Notes provide additional information or suggestions about the topic or procedure being discussed.

Read text marked by the "Important" icon for information that Important will help you complete a procedure or avoid extra steps.



Read text marked by the "CAUTION" icon for information you Caution must know to avoid damaging the RDX QuikStation or losing data.



Read text marked by the "WARNING" icon for information you must know to avoid Warning personal injury.

Х

Image: Second state PRODUCT OVERVIEW

The RDX® QuikStation[™] is a multi-drive, network-attached, removable disk-based library built on RDX technology, which combines the removability, durability and economy of tape with the random accessibility and performance of disk. Depending on the size of the RDX cartridges that are in use, the RDX QuikStation can offer up to 8-terabytes of online capacity.

This chapter provides an overview of the RDX QuikStation's features and components.



RDX QUIKSTATION FEATURES

This section provides an overview of the RDX QuikStation's features.

CONFIGURATION FLEXIBILITY

You can configure the RDX QuikStation as one of the following device types:

- Eight Tandberg Data RDX drives with eight removable iSCSI disk targets.
- Eight LTO-3 tape drive iSCSI targets.
- A disk autoloader with one RDX drive and eight slots.
- A Tandberg Data StorageLibrary T24 with two LTO-3 tape drives and eight slots.
- A Tandberg Data StorageLoader LTO with one LTO-3 tape drive and eight slots.
- A Tandberg Data StorageLibrary T40+ with four LTO-3 tape drives and eight slots.
- A tape autoloader with one LTO-3 tape drive and eight slots.

For even greater flexibility, you can configure the following hybrid device types:

- A virtual Tandberg Data StorageLibrary T24 with two LTO-3 tape drives and four slots plus four RDX drive targets.
- A virtual Tandberg Data StorageLoader LTO with one LTO-3 tape drive and four slots plus four RDX drive targets.

TANDBERG DATA RDX REMOVABLE DISK TECHNOLOGY

RDX data cartridges combine the removability and durability of tape with the random access and performance of disk. RDX cartridges have up to a 30-year media shelf life, and a shock-proof and static-proof cartridge design that makes them ideal for data archiving.

HIGH PERFORMANCE AND CAPACITY

When configured as eight removable disks, the RDX QuikStation can simultaneously read and write cartridges from all eight RDX docks, up to 200 megabytes per second¹. When configured as a virtual Tandberg Data StorageLibrary T24, the RDX QuikStation can simultaneously read and write from both tape drives.

With all eight docks using 1-terabyte cartridges, the RDX QuikStation has an online capacity of 8-terabytes, and unlimited offline storage capacity.

COMPATIBLE WITH MAJOR BACKUP SOFTWARE

The RDX QuikStation is compatible with major backup software applications, such as CA ARCserve® and Symantec Backup Exec.[™] For a complete list of backup software applications that are compatible with the RDX QuikStation, visit the Tandberg Data web site (www.tandbergdata.com), or contact your backup software vendor.

INTEGRATED WEB-BASED MANAGEMENT

The RDX QuikStation Remote Management Console allows you to use a standard web browser from any location to manage, control, diagnose, and configure security for the RDX QuikStation.

DUAL GIGABIT ETHERNET

The RDX QuikStation has two Gigabit Ethernet ports to provide failover protection and high availability network connectivity. You can configure the ports for either active backup/failover or adaptive load balancing (see page 51).

COMPACT DESIGN

The RDX QuikStation uses only two units (2U) of rack space, minimizing the impact on data center space requirements.

^{1.} The data transfer rate is up to 200 MB/second when port configuration is configured for adaptive load balancing (see page 51) and multiple hosts are used to stream data.

RDX QUIKSTATION COMPONENTS

This section describes the RDX QuikStation's front panel and back panel components.



Figure 1-1 RDX QuikStation front panel components

| 0 | RDX LCD panel (eight each) — see page 6 for more information. |
|---|--|
| 2 | Emergency-eject keyhole (eight each). Note: To use the emergency-eject function, straighten a paper clip (or use a similar straight metal object) and insert it into the keyhole. The paper clip needs to be sturdy enough to engage the eject function. |
| 8 | RDX dock (eight each). |
| 4 | RDX cartridge-eject button (eight each) — see page 7 for more information. |
| 6 | USB 2.0 port — used for configuration and diagnostics. |
| 6 | LED panel — displays indicators for network, USB connection, and system health (see page 7 for more information). |



Figure 1-2 RDX QuikStation back panel components

| 0 | Power connector |
|---|---|
| 2 | Power button |
| | Note: When the unit is turned on, pressing the power button performs a soft shutdown (recommended). Pressing and holding the power button for 10 seconds performs a hard shutdown. |
| 3 | Mouse connector (green) |
| 4 | Keyboard connector (purple) |
| 6 | USB 2.0 port (two each) |
| 6 | Serial connector |
| 7 | VGA connector |
| 8 | Gigabit Ethernet port (two each) |

RDX DOCK LCD PANELS

The eight RDX dock LCDs display information based on the conditions that are described in the following table

| Condition | What is displayed in LCD Dock Panels | |
|---|---|--|
| Upon initial startup of the unit and before a logical device type has been selected. | The firmware level, system name, IP status (DHCP or static), and a valid IP address (if DHCP is enabled) continuously display until you select a logical device type from the Remote Management Console. | |
| After a system restart. | The firmware level, system name, IP status (DHCP or static), and IP address display in sequential order for a few seconds. For RDX docks that have a loaded cartridge, the cartridge label number displays in the LCD panel. | |
| When a USB flash drive is inserted into | "LogDump" displays. When the system log dump is complete, the LCD panels display "PASSED." | |
| the front-panel USB | If one of the follow | ving files is loaded on the USB stick: |
| | network.conf: | "NetUpdate" displays. When the configuration download is complete, the LCD panels display "PASSED." |
| | password.reset: | "RstPasswd" displays. When the password reset is complete, the LCD panels flash "DONE." |
| | vtx.fw: | "CodeUpdate" displays. When the firmware update is complete, the LCD panels display "Remove stick to reboot." |
| An RDX cartridge is loaded in a dock. | When the dock is configured as eight removable disks, the LCD panel displays the last eight digits of the cartridge serial number. When the dock is configured as a tape library slot, the LCD panel displays an LTO-3 "tape label", which consists of the last six digits of the cartridge serial number followed by "L3." | |
| An RDX dock is empty. | Nothing is displayed. | |

RDX CARTRIDGE-EJECT BUTTON/POWER LED

The following table describes Eject Button/Power LED functionality.

| LED Color | Status |
|----------------|--|
| Off (no color) | The unit does not have power. |
| Green | Power is on and the unit is working properly. |
| Blinking Green | The RDX dock is ejecting a cartridge. |
| Amber | A fault condition exists with the cartridge. |
| Blinking Amber | A host computer is accessing the cartridge. A user has either pressed the eject button or has ejected the cartridge via the Remote Management Console. |

RDX QUIKSTATION LED INDICATOR PANEL

The following table describes LED indicator functionality on the front panel.

| LED Indicator | Color | Status |
|------------------|-------------------|--|
| 문급 | Off (no color) | RDX QuikStation is not connected to a network cable. |
| 윰 | Blinking Green | Network connection has good-to-excellent connectivity. |
| 윰 | Blinking Amber | Network connection has limited or no connectivity. |
| • | Blinking Green | System is loading information from or writing data to the USB flash drive. |
| • | Solid Green | USB flash drive is ready to be removed. |
| • <u>/</u> • | Off (no color) | No USB device is detected. |
| | Off (no color) | No system faults exist. |
| | Blinking Amber | A fault exists. The fault could be a fan failure, or it could indicate the device temperature is too high. |

8

2

INSTALLATION AND SETUP

This chapter describes how to install and configure the RDX QuikStation and includes the following sections:

| Preparing for Installation page 10 |
|---|
| Installing the RDX QuikStation in a Rack page 12 |
| Connecting Network and Power Cables page 17 |
| Configuring Network Information page 18 |
| Connecting to the Remote Management Console page 20 |
| Configuring the RDX QuikStation page 21 |

PREPARING FOR INSTALLATION

✓ For all host computers that will be backed up by the RDX QuikStation, verify that the host computer OS has iSCSI initiator software installed.

| Windows | Linux | Macintosh |
|--|---|---|
| iSCSI initiators are included with Windows Server 2008 and later. If necessary, download and install the appropriate iSCSI initiator from the Microsoft web site or | iSCSI initiators are included with most Linux operating systems. If necessary, download and install the appropriate iSCSI initiator from your Linux Distribution web site or | Macintosh systems require the installation of a third-party iSCSI initiator. We recommend that you download and install an iSCSI initiator that has been tested or recommended for use |
| from a third-party | from a third-party | with your chosen backup |
| vendor. | vendor. | application. |

- ✓ Verify all items are included in the accessory kit (see page 11).
- ✔ Keep the packing materials and box in case you need to ship the unit later.
- ✓ Select an appropriate location —The maximum recommended ambient temperature for the RDX QuikStation is 35° C (95° F). Install the appliance in an environment that is compatible with this temperature limit. Use a ventilated rack to ensure the appliance receives adequate air flow required for safe operation.
- ✓ Select an appropriate power source The AC input for the RDX QuikStation is 100-240 VAC, and the rated input current is 2,0-1,0A. Keep this information in mind when selecting a power source to prevent circuit overload. Plug the appliance's power cord into a high-quality power strip that offers protection from electrical noise and power surges. An uninterruptible power supply (UPS) is recommended.
- ✓ Ensure that the work area is free from conditions that could cause electrostatic discharge (ESD) — Discharge static electricity from your body by touching a known grounded surface, such as a computer's metal chassis.
- ✓ Ensure the rack is placed on level ground Use a level when installing the rack rails to prevent any potentially hazardous conditions caused by uneven mechanical loading.
- ✔ Obtain the following items to complete the rack installation instructions
 - ▶ #2 Phillips screwdriver
 - Tape measure
 - Level

VERIFYING ACCESSORY KIT CONTENTS

Use the following checklist to verify that all accessory items shipped with the RDX QuikStation:

| ~ | RDX QuikStation Accessory Item |
|---|--|
| | North American-region power cord |
| | European-region power cord |
| | Two ethernet cables |
| | Two appliance end-cap covers to place over rackmount bolts |
| | USB flash drive |
| | Rackmount rail kit, which includes: |
| | Two inner rack rails to attach to the RDX QuikStation |
| | Two outer rack rails to attach to the rack |
| | Hardware pack for inner rails |
| | Hardware pack for outer rails |
| | Rail extension brackets |
| | Eight washers and two clip nuts used for rackmounting |
| | RDX QuikStation Quick Start Guide |
| | RDX QuikStation Rackmount Instructions |
| | SecureService Pack, which includes: |
| | Warranty card |
| | SecureService registration card |
| | Service and support information |
| | Data Protection Solutions Guide |

 Table 2-1
 RDX QuikStation accessory kit contents

INSTALLING THE RDX QUIKSTATION IN A RACK

The RDX QuikStation is designed for installation in a standard (EIA-310) 4-post, 19-inch rack and uses 2 units (2U) of rack space.

The rackmount kit includes the following items:

Table 2-2Rackmount kit contents

| Rackmount Item | | What the item is used for |
|--|---------------------------|--|
| Ĩ | 12 #10 pan-head screws | To attach the bracket to the outer rail and to secure the outer rail to the front and back-end of the rack. |
| | 4 #10 keps nuts | These are extra hardware pieces that are not required for assembly. |
| | 4 #10 flat-head screws | These are extra hardware pieces that are not required for assembly. |
| 6 | 8 washers | To use with the #10 pan-head screws when you attach the outer rail to the rack. |
| 0 0 0 | 6 bar nuts | To secure the outer rail to the rack and also to secure the bracket to the outer rail. |
| | 2 rail extension brackets | To attach the outer rail to the back-end of the rack. |
| 0 0 | 2 short brackets | To attach the outer rail to the back-end of the rack if the rack does not require rail extension brackets. |
| | 2 outer rails | To mount to the rack. |
| | 2 inner rails | To attach to the RDX QuikStation appliance. Includes a locking tab to keep the appliance in place when it is fully extended from the rack. |
| Ŷ | 8 #6 pan-head screws | To secure the inner rail to the RDX QuikStation. |
| | 8 #6 keps nuts | These are extra hardware pieces that are not required for assembly. |
| and the second sec | 2 #10 clip nuts | To secure the RDX QuikStation into the rack. |

ASSEMBLING THE OUTER RAILS

For most standard, 4-post racks, you will need to attach the rail extension bracket (shown below) to the outer rail instead of the short bracket.



Figure 2-1 RDX QuikStation outer rack rail assembly

To assemble the outer rails:

- 1. Measure the inside distance from the front mounting holes to the back mounting holes on one side of your rack.
- 2. Fit either the rail extension bracket or the short bracket to the outer rail. Adjust the bracket to a point where the rail is at the same distance that you measured in Step 1.
- 3. Align the bottom and middle hole of a bar nut to the outside of the bracket, as shown in the above figure. From the inside of the rail, thread two #10 pan-head screws through the rail, bracket and bar nut holes, as shown above. Do not tighten the screws completely.
- 4. Repeat these steps for the second outer rail on the other side of the rack.

INSTALLING THE OUTER RAILS INTO THE RACK

The front and back of the rails attach to the rack with a bar nut, two washers, and two #10 pan-head screws, as shown in the figure below.



Figure 2-2 Attaching the outer rail to the rack

Securing the front of the rail to the rack

- 1. Place an outer rail against the *inside* flange of the rack post (see figure above). Make sure you orient the rail so that the bracket (either the rail extension or the short bracket) faces the back of the rack.
- 2. Place a bar nut flush against the end of the rail (see figure above). Position the rail and bar nut so that the bar nut's bottom hole and middle hole align with rack holes that are 0.5" apart (see figure above).
- **3.** Thread the screws through the washers, the rack post, and the bottom and middle holes of the bar nut (see figure above).
- 4. Install a clip nut in the hole directly below the bar nut (see figure above).

Securing the back of the rail to the rack

- 1. Line up the rail with the back placement holes in the rack. Make sure the rail is against the inside rack flange and is level to the ground.
- 2. Thread the screws through the washers, the rack post, and the bottom and middle holes of the bar nut (see figure above).
- **3.** If necessary, adjust the rail so that it fits snugly into the rack. Tighten the two #10 pan-head screws that attach the bracket to the outer rail.

Repeat these steps to attach the other outer rail and clip nut to the rack.

ATTACHING THE INNER RAILS



Figure 2-3 Attaching the inner rails to the RDX QuikStation

- 1. On each side of the appliance, align the inner rail holes with the first three holes on the RDX QuikStation.
- 2. Use three #6 pan-head screws to attach the inner rail to the appliance, as shown in the above figure.

Be sure to use the supplied #6 screws – a longer screw could cause damage to the unit.



INSTALLING THE RDX QUIKSTATION INTO THE RACK

Figure 2-4 Installing the RDX QuikStation into the rack

To install the RDX QuikStation into the rack:

- 1. Line up the rear of the inner rails with the front of the outer rails.
- 2. Slide the inner rails into the outer rails, keeping the pressure even on both sides. You may have to depress the locking tabs when inserting.
- 3. Push the RDX QuikStation completely into the rack.
- 4. Use the locking screws located on each side of the RDX QuikStation to secure it into the rack. Push and turn the locking screws clockwise to lock. The locking screws engage the clip nut, which you installed in Step 2d of "Installing the Outer Rails into the Rack" on page 14.
- 5. Place the appliance end-cap covers that were included in your accessory kit over the locking screws.

Important Do not place anything on top of the rackmounted unit when it is extended from the rack.

CONNECTING NETWORK AND POWER CABLES



Figure 2-5 Connecting ethernet and power cables to the RDX QuikStation

- 1. Connect one or two RDX QuikStation ethernet network ports to ethernet network ports on a switch, hub, or router.
 - **Note:** Two network ports are recommended for failover protection. (See page 51 for more information about port configuration options.)
- 2. Connect the power cord that is appropriate for your region to the RDX QuikStation.
- **3.** Turn the power on by pressing the black power button that is located on the back of the RDX QuikStation.
 - **Note:** Once the unit is powered on, all eight LCD panels will alternately display the firmware level, system name, DHCP status, and IP address. This information continues to display until you log on to the Remote Management Console and configure the logical device type for the QuikStation.
- 4. Verify that a valid IP address is displayed on the LCD panels. If not, you must manually configure the IP address for the RDX QuikStation. (See "Configuring Network Information" on page 18.)

CONFIGURING NETWORK INFORMATION

Note: If you are able to obtain a valid IP address from your RDX QuikStation LCD panels upon startup, skip this section and go to Connecting to the Remote Management Console on page 20.

To manually configure your RDX QuikStation network information:

- 1. Locate the USB flash drive that is included in your RDX QuikStation accessory box and insert the flash drive into an open USB slot on any computer with a text editor.
- 2. From the computer, navigate to the USB Flash Drive. Open the **network.conf** file in a text editor. (For Windows users, open the file in Notepad.)
 - **Note:** If you cannot find the USB flash drive that shipped with the RDX QuikStation, you can download the **network.conf** file from the **Support>Drivers and Downloads** section of the Tandberg Data web site (www.tandbergdata.com).

| network.conf - Notepad |
|--|
| File Edit Format View Help |
| # Tandberg Data RDX QuikStation Network Configuration File: |
| # # Use this file to manually configure the RDX QuikStation IP address to # IPv4 address via a USB flash drive. NOTE: If automatic IP address as # via DHCP is desired, no action is necessary as DHCP is the factory de |
| # # Upon startup, the QuikStation will display the current IP settings an # across all LCD's. However, following initial setup and configuration # Data recommends that you assign a static IP address to the RDX QuikSt # prevent inadvertent IP address changes of the iSCSI target. # |
| # Follow the steps below to configure a static IPv4 address: |
| <pre># # Step 1: Obtain an available IP address and subnet mask for the RDX Qu # Edit the following configuration statements to meet your network's re # Since any line that begins with '#' is ignored by the RDX QuikStation # the '#' from each modified line so the QuikStation can decipher the s # changes. Note that if you choose to specify a static IP address, bot # SUBNETMASK fields must be "uncloaked" (GATEWAY is optional).</pre> |
| <pre># # start edit section # IP=192.168.10.2 # SUBNETMASK=255.255.0 # GATEWAY=192.168.10.1 # end edit section</pre> <pre> Edit this section </pre> |
| # Step 2: Save this newly-edited file to the root directory of the USB # stick included with the RDX QuikStation. If the included USB stick i # any USB memory stick with a VFAT file system will suffice. IMPORTANT # of this file must remain network.conf for the RDX QuikStation to reco # process it. |
| # Step 3: Insert the USB stick into the USB port located in the center # front panel of the RDX QuikStation. The USB symbol on the center-fron # will flash and the LCD's will broadcast the message "NetUpdate". Whe # process is complete, the LCD's will flash the message "PASSED" and th # will stop flashing and illuminate solid indicating that it is safe to # USB stick. Remove the USB stick. # |



3. In the **network.conf** file, edit the IP and subnet mask addresses to meet your network's requirements.

Note: You can edit the gateway address, but it is not required.

4. Remove the "#" character from each modified line so that the RDX QuikStation can decipher the setting changes, as shown in the example below.

Step 1: Obtain an available IP address and subnet mask for the RDX QuikS # Edit the following configuration statements to meet your network's requi # Since any line that begins with '#' is ignored by the RDX QuikStation, r # the '#' from each modified line so the QuikStation can decipher the sett # changes. Note that if you choose to specify a static IP address, both I # SUBNETMASK fields must be "uncloaked" (GATEWAY is optional).

```
# -- start edit section --
IP=10.1.20.85
SUBNETMASK=255.255.255.0
GATEWAY=10.1.20.254
# -- end edit section --
```

Step 2: Save this newly-edited file to the root directory of the USB mem # stick included with the RDX QuikStation. If the included USB stick is u # any USB memory stick with a VFAT file system will suffice. IMPORTANT: T # of this file must remain network.conf for the RDX QuikStation to recogni # process it. # # Step 3: Insert the USB stick into the USB port located in the center of

Step 3: Insert the USB stick into the USB port located in the center of # front panel of the RDX QuikStation. The USB symbol on the center-front L # will flash and the LCD's will broadcast the message "NetUpdate". When t # process is complete the LCD's will flash the message "PASSED" and the U Figure 2.7 Example of an edited network confile

Figure 2-7 Example of an edited network.conf file

5. Save the file to the root directory of the USB flash drive as "**network.conf**." If the flash drive that was included with your RDX QuikStation is unavailable, you can save the file to any USB flash drive with a VFAT file system.

Important The file name must remain as "network.conf." Otherwise, the RDX QuikStation will not recognize the file.

6. Insert the USB flash drive into the USB port located on the center-front panel of the RDX QuikStation.

After you insert the flash drive, "Net Update" will display on all eight front LCD panels. When the configuration download is complete, the LCD panels display "PASSED."

- 7. When the USB symbol on the center-front LED panel stops blinking and transitions to a solid green color, remove the USB flash drive.
- 8. Proceed to the next section to connect to the Remote Management Console.

CONNECTING TO THE REMOTE MANAGEMENT CONSOLE

You can connect to the RDX QuikStation Remote Management Console with a web browser from any computer that is on the same network as the RDX QuikStation. The computer can run on Windows, Linux, and Macintosh operating systems. You can also enable secure hypertext (https) when connecting to the Remote Management Console.

Note: Your web browser must have JavaScript enabled.

To connect to the RDX QuikStation Remote Management Console:

- 1. From a network host, type the RDX QuikStation IP address in a web browser: http://<*RDX QuikStation IP address*>
- **Note:** If your network runs DHCP, you can obtain the IP address from the RDX QuikStation's LCD panels.
- 2. Sign in to the Console. The default user name is Admin, and the default password is Admin!.

Note: The default password is case sensitive.

| Sign In to the RDX QuikStation ? | | | |
|----------------------------------|---------------------|-------|--|
| User Name: | Admin | | |
| Password: | | | |
| The defa | ult password is Adm | in! | |
| | | | |
| | Sign | un du | |

The Remote Management Console opens.



3. Go to the next section to configure your RDX QuikStation logical device type and other settings.

CONFIGURING THE RDX QUIKSTATION

LOGICAL DEVICE TYPE

You can configure your RDX QuikStation in a variety of ways. Logical device configurations are available for disk, tape, disk automation and tape automation. For even greater flexibility, you can choose a hybrid configuration of virtual tape library and disk.

For more information about best use for each logical device type, see Chapter 5.

To configure the logical device type:

- 1. Select the device type from the dialog window that opens when you first log on to the Remote Management Console.
- **Note:** If you select the RDX Drive configuration, you must first install the RDX Utility, which is available from a link in the Logical Device Type Settings dialog window. The RDX drive is the recommended device type for disk configurations. The removable disk device type is available for legacy users.

| This action will eject all the cartridges in the RDX QuiKStation and res about logical device types, refer to the RDX QuiKStation product mai If you are using a device type with an RDX drive, such as t must install RDX Utility on each Windows host attached to eject button will not work properly. Download the latest RDX Utility <u>Here.</u> | start the system. For more information nual. he Disk Configuration option, you an RDX device. Otherwise, the |
|--|---|
| - Current Selected Type TD StorageLibrary T24 (Two LTD-3 drives and eight | slots) |
| Disk Configuration | + |
| Tape Configuration | + |
| Disk Automation Configuration | + |
| Tape Automation Configuration | E |
| TD StorageLibrary T24 (Two LTO-3 drives and eight slots) TD StorageLoader LTO (One LTO-3 drive and eight slots) TD StorageLibrary T40+ (Four LTO-3 drives and eight slots) Tape Autoloader (One LTO-3 drive and eight slots) | |
| Hybrid Configuration | + |
| | |

2.Click**OK**.

SYSTEM SETTINGS

Refer to the table below to configure the following RDX QuikStation system settings. For more information, see Chapter 4 on page 47.

| Table 2-3 | Recommended | system | settings |
|-----------|-------------|--------|----------|
|-----------|-------------|--------|----------|

| Setting | Purpose | Where to find |
|------------------------|---|---|
| Date/Time | Set the date and time to your time zone to receive accurate system and troubleshooting information. See "Setting Date and Time" on page 49 for more information. | From the main menu bar in the Remote Management Console, go to System Settings> Date/Time. |
| Network Information | Tandberg Data recommends that you assign a static IP address to prevent inadvertent IP address changes of the iSCSI target. You can also configure optional DNS server addressing information. See "Configuring Network Information" on page 50 for more information. | From the main menu bar in the Remote Management Console, go to System Settings> Network. The RDX QuikStation will automatically reboot after you have made your network changes. |
| Email Notification | A valid email address is needed for logging, error reporting, and password recovery. See "Setting Email Notifications" on page 53 for more information. | From the main menu bar in the Remote Management Console, go to System Settings> Notification. |
| User Management | As a best practice, change the built-in administrator account password. See "Editing a password" on page 63 for more information. | From the main menu bar in the Remote Management Console, go to System Settings>User. |

3

ISCSI CONFIGURATION

The data that you want to back up and restore is transferred from your host computers to the RDX QuikStation via the iSCSI (Internet Small Computer System Interface) protocol. In iSCSI terminology, a host computer is referred to as an "initiator" and the storage device it connects to as the iSCSI "target."

This chapter includes the following sections:

| ▶ iSCSI configuration overview | age 24 |
|--|--------|
| • Connecting to iSCSI targets without authentication | age 25 |
| Setting iSCSI security | age 33 |

ISCSI CONFIGURATION OVERVIEW

Before you can begin to back up and restore data with the RDX QuikStation, the host computers (initiators) must establish an iSCSI connection to the RDX QuikStation's storage targets.

To establish an iSCSI connection, you need to have iSCSI initiator software installed on the host. Verify that the host computer OS has iSCSI initiator software installed (see page 10).

Depending on how you configured the logical device type for the RDX QuikStation in Chapter 2 (see page 21), you will have two to eight iSCSI targets to connect.

Note: For more information about viewing iSCSI targets, see "iSCSI Device Overview" on page 72.

You can configure iSCSI initiators with or without iSCSI security. The RDX QuikStation supports one-way Challenge Handshake Authentication Protocol (CHAP). CHAP is a protocol that is used to authenticate iSCSI connections and is based upon the initiator and target sharing a *secret* (a security key that is similar to a password).

With one-way CHAP authentication, the iSCSI target (storage device) authenticates the initiator (host). The secret is set just for the target. All initiators that want to access that target need to use the same secret to connect to the target. For more information, see "Setting iSCSI Security" on page 33.

The following table describes iSCSI Management rules for the RDX QuikStation.

| Logical Device type | iSCSI Management Security | iSCSI Management Connection |
|--|--|--|
| Eight RDX drives ^a | You can add up to 10 initiator users | |
| Eight tape drives | (CHAP authentication) or hosts (host authentication) to each iSCSI target. | |
| Disk autoloader | You can add up to 10 initiator users | |
| StorageLibrary T24 with two LTO-3 tape drives | (CHAP authentication) or hosts (host authentication) to the library or autoloader target. You cannot set security on the drive target; the drive inherits security from the library or autoloader. | One active host connection is allowed for each iSCSI target. |
| StorageLoader with one LTO-3 tape drive | | |
| StorageLibrary T40+ with four LTO-3 tape drives | , , | |
| Tape autoloader with one LTO-3 tape drive | | |
| Hybrid Configurations | For hybrid configurations, refer to iSCSI management rules for the given logical device types. | |

 Table 3-1
 Remote Management Console iSCSI Management Rules

^a The RDX drive is the recommended device type for disk configurations. The removable disk device type is available for legacy users.

CONNECTING TO ISCSI TARGETS WITHOUT AUTHENTICATION

This section describes how to connect Windows, Linux, and Macintosh hosts to RDX QuikStation iSCSI targets without authentication. If you plan on using one-way CHAP authentication, follow the instructions in "Setting iSCSI Security" on page 33.

- Connecting iSCSI Targets for Windows (page 25)
- Connecting iSCSI Targets for Linux (page 28)
- Connecting iSCSI Targets for Macintosh (page 30)

CONNECTING ISCSI TARGETS FOR WINDOWS

The following instructions describe how to connect your network host computers in Windows Server 2008 R2. Your iSCSI initiator interface may differ slightly depending on your Windows operating system.

To establish an iSCSI connection:

- 1. Open Microsoft iSCSI Initiator and click the Discovery tab.
- 2. Click Discover Portal...). (Your version of iSCSI initiator may use Add depending on your Windows OS.)

| o add a target portal, click Discover Portal. o remove a target portal, select the address above and nen click Remove. | els Discovery | Favorite Targe | ets Volumes and Devices | RADIUS Configuratio |
|---|-------------------------------------|-------------------|-------------------------|---------------------|
| Address Port Adapter IP address o add a target portal, click Discover Portal. Discover Portal. Discover Portal. o remove a target portal, select the address above and nen click Remove. Remove | arget portals The system will lo | ook for Targets o | n following portals: | Refresh |
| add a target portal, click Discover Portal. remove a target portal, select the address above and Remove n click Remove. | ldress | Port | Adapter | IP address |
| | | | | |

The Discover Target Portal dialog window opens.

| Discover Target Portal | ••• |
|---|------------------------------------|
| Enter the IP address or DNS name and p want to add. | port number of the portal you |
| To change the default settings of the dis the Advanced button. | covery of the target portal, click |
| IP address or DNS name: | Port: (Default is 3260.) |
| 10.1.50.211 | 3260 |
| Advanced | OK Cancel |

- 3. In the Discover Target Portal dialog window, type the IP address for the RDX QuikStation. Click OK.
- 4. Click the Targets tab.

Depending on how you configured the logical device type for the QuikStation, you will see two to eight iSCSI targets. The screenshot below depicts a TandbergData StorageLibrary T24 configuration. The three iSCSi targets connect the library and two LTO-3 tape drives to the host.

| CSI Initiator Properties | |
|---|----------------------|
| Targets Discovery Favorite Targets Volumes and Devices | RADIUS Configuration |
| Ouick Connect | |
| To discover and log on to a target using a basic connection, type the IP address or DNS name of the target and then click Quick Connect. | |
| Target: | Quick Connect |
| Discovered targets | |
| | Refresh |
| Name | Status |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 | Inactive |
| | |
| To connect using advanced options, select a target and the click Connect. | Connect |
| To completely disconnect a target, select the target and then click Disconnect. | Disconnect |
| For target properties, including configuration of sessions, select the target and click Properties. | Properties |
| For configuration of devices associated with a target, select the target and then click Devices. | Devices |
| More about basic ISCSI connections and targets | |
- **5.** For each iSCSI target, complete the following steps:
 - a. Select an iSCSI qualified name (IQN). Click Connect. (Your version of iSCSI initiator may use LogOn...) depending on your Windows OS.)

The Connect to Target dialog window opens.

| Connect To Target | × |
|--|---|
| Target name: | |
| 10-01.com.tandbergdata:storage.MAGNUM | 1-224-1.002590089E6A |
| Add this connection to the list of Favorite This will make the system automatically a connection every time this computer rest | e Targets. ttempt to restore the earts. |
| 🔲 Enable multi-path | |
| Advanced | OK Cancel |

b. Select the "Add this connection to the list of Favorite Targets" checkbox. Click OK.

The RDX QuikStation's iSCSI target should display as "connected" in the Discovered Targets dialog window, as shown below.

| argets Discovery | Favorite Targets | Volumes and Devices | RADIUS | Configuration |
|---|---|--|---------------------|---------------|
| Quick Connect | 1 | | | |
| To discover and lo DNS name of the t | g on to a target usin arget and then click | ig a basic connection, t Quick Connect. | ype the IP | address or |
| Target: | | | Qu | iick Connect |
| - | | | | |
| - Discovered targets | 5 | | | |
| Discovered targets | 5 |] | | Refresh |
| Discovered target: | 5 | | Status | Refresh |
| Discovered targets | ,tandbergdata:stora | ige.MAGNUM-224 | Status | Refresh |
| Discovered targets Name iqn.2010-01.com | ; .tandbergdata:stora .tandbergdata:stora | ige.MAGNUM-224 ige.MAGNUM-224 | Status Connected | Refresh |

Important Before you shutdown or restart the RDX QuikStation for any reason, Tandberg Data recommends that you disconnect all iSCSI initiators.

CONNECTING ISCSI TARGETS FOR LINUX

The following instructions describe how to connect iSCSI targets using Linux Open-iSCSI Initiator. Open-iSCSI Initiator is available for Redhat Enterprise Linux, CentOS, Fedora Linux, Debian, and Ubuntu operating systems. For more information about using Open-iSCSI with your specific Linux operating system, refer to your Linux distribution web site.

1. Install Open-iSCSI initiator.

To install the iSCSI initiator, enter the following command:

yum install iscsi-initiator-utils

For Ubuntu and Debian operating systems, enter the following command:

\$ sudo apt-get install open-iscsi

2. Discover iSCSI Targets.

To discover the RDX QuikStation iSCSI targets, enter the following command, replacing the variable *RDX QuikStation IP address* with the actual IP address for your RDX QuikStation:

```
# iscsiadm -m discovery -t sendtargets -p RDX QuikStation
ip address
```

Depending on how you configured the logical device type for the QuikStation, you will see two to eight iSCSI targets.

3. Connect iSCSI targets.

To connect all discovered iSCSI targets, enter the following command:

iscsiadm -m node -L all

To connect to an individual target, enter the following command, replacing *target IQN name* with the iSCSI qualified name (IQN) of the target and replacing *target portal* with the RDX QuikStation IP Address:

iscsiadm -m node -T target IQN name -p target portal -- login

Once all iSCSI targets are connected, the initiator (Linux host) can use the iSCSI disks as a normal disk. It appears under/dev/sdx devices and you can format, mount iSCSI disks like a normal disk.

Other "iscsiadm" commands you may need

1. Disconnect from an iSCSI target.

To disconnect from an iSCSI target, enter the following command, replacing *target IQN name* with the iSCSI qualified name (IQN) of the target and replacing *target portal* with the RDX QuikStation IP Address:

```
# iscsiadm -m node -T target IQN name -p target portal --logout
```

2. Delete an iSCSI target.

To delete a target, enter the following command, replacing *target IQN name* with the iSCSI qualified name (IQN) of the target:

```
# iscsiadm -m node -T target IQN name -o delete
```

Important Before you shutdown or restart the RDX QuikStation for any reason, Tandberg Data recommends that you disconnect all iSCSI initiators.

CONNECTING ISCSI TARGETS FOR MACINTOSH

The following instructions describe how to connect your Macintosh host computers using Atto Technology's Xtend SAN iSCSI initiator. Xtend SAN is compatible with Mac OS X 10.4.x and later. While we do not require you to use Xtend SAN as your third-party Mac iSCSI initiator, we recommend that you use an iSCSI initiator that has been tested or recommended for use with your chosen backup application.

To establish an iSCSI connection:

1. Open Xtend SAN iSCSI Initiator. Click the Discover Targets tab, then click Discoverby DNS/IP.

| 0 0 | ATTO iSCSI Mac InitiatorXtend SAN |
|-------------------------------------|--|
| Hosts Tandberg-Data-ISVs-MacBook | Initiator Discover Targets |
| | Discover Targets Configure the static discovery. |
| | Address: 10.1.50.211 |
| | Port: 3260 |
| | СНАР |
| | Target User Name: iqn.1995-12.com.attotech:xtendsan:ser.wq9E |
| • | Target Secret: |
| | Mutual Authentication: |
| | Initiator User Name: |
| | Initiator Secret: |
| | |
| | Finish Cancel |

The following dialog window opens.

2. In the Address field, type the IP address for the RDX QuikStation. If you are using a port number other than the default, type the number in the Port field. Otherwise leave the default 3260 port number. Click Finish.

The dialog window displays the discovered iSCSI targets. Depending on how you configured the logical device type for the QuikStation, you will see two to eight iSCSI targets. The screenshot below depicts a configuration of eight removable disks.

| 000 | ATTO iSCSI Mac InitiatorXtend SAN |
|-------------------------------------|--|
| Hosts Tandberg-Data-ISVs-MacBook | Ar to ises inde initiatorxtend sAN Initiator Discover Targets Discovered Targets ign.2010-01.com.tandbergdata:storage.rdx8-8.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx7-7.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx4-6.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx4-4.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx3-3.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx3-3.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a ign.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a |
| | Add Clear |

3. Highlight all the discovered targets and click Add.

The targets are displayed in the left pane with a red icon, signifying that they are not yet connected.

| \varTheta 🔿 🔿 🔿 ATTO iSCSI Mac Initia | atorXtend SAN |
|---|---|
| Hosts ▼ ■ Tandberg-Data-ISVs-MacBook-Pro.local ● iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx3-3.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx4-4.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx5-5.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx6-6.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx7-7.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089 | Initiator Discover Targets Discover by DNS/IP Discover by iSNS Discovered Targets iqn.2010-01.com.tandbergdata:storage.rdx8-8.0025 iqn.2010-01.com.tandbergdata:storage.rdx6-6.0025 iqn.2010-01.com.tandbergdata:storage.rdx4-4.0025 iqn.2010-01.com.tandbergdata:storage.rdx3-3.0025 iqn.2010-01.com.tandbergdata:storage.rdx2-2.0025 iqn.2010-01.com.tandbergdata:storage.rdx1-1.0025 |
| ● iqn.2010-01.com.tandbergdata:storage.rdx7-7.00259008 ● iqn.2010-01.com.tandbergdata:storage.rdx8-8.00259008 | <pre>iqn.2010-01.com.tandbergdata:storage.rdx5-5.00, iqn.2010-01.com.tandbergdata:storage.rdx4-4.00; iqn.2010-01.com.tandbergdata:storage.rdx3-3.00; iqn.2010-01.com.tandbergdata:storage.rdx2-2.00; iqn.2010-01.com.tandbergdata:storage.rdx1-1.00; Add Clear</pre> |

- 4. For each iSCSI target, complete the following steps to connect the target:
 - a. Select an iSCSI qualified name (IQN) from the left pane.

The target's port information opens in the center pane.

| Setup Status LUNs | | | | |
|--|---|--|--|--|
| Name: iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a | | | | |
| | | | | |
| | | | | |
| Visible | Auto Login | | | |
| ✓ | \checkmark | | | |
| | Setup Status LUNs .tandbergdata:storage.rds Visible | | | |

- **b.** Select the "Visible" checkbox and the "Auto Login" checkbox.
- c. Click the Status tab.

The target's connection status is displayed.

| | Setup Status | LUNs |
|-----------------------|-----------------------|------------------------|
| Name: iqn.2010-01.co | m.tandbergdata:storag | ge.rdx1-1.002590089e6a |
| Alias: | | |
| | | |
| Ports | | |
| Ports Network Node | Status | Auto Login |

- d. Select the Network Node and click Login.
- e. The iSCSI targets will display with a green icon when they are connected to the RDX QuikStation, as shown below.



Important Before you shutdown or restart the RDX QuikStation for any reason, Tandberg Data recommends that you disconnect all iSCSI initiators.

SETTING ISCSI SECURITY

The RDX QuikStation allows you to set iSCSI security in one of the following ways

| Security Type | Description |
|------------------------|--|
| CHAP Authentication | CHAP is a protocol that is used to authenticate iSCSI connections and is based upon the initiator and target sharing a <i>secret</i> (a security key that is similar to a password). You can add up to 10 initiator users (hosts) per iSCSI target ^a . However, for removable disk configurations, only one initiator can be actively connected to the disk target. For virtual library configurations, only one initiator can be actively connected to a logical tape drive. |
| Host Authentication | Only the hosts that you specify for an iSCSI target will be allowed to connect to the target. Without host authentication, any network host can connect to the iSCSI target. You can add up to 10 hosts per iSCSI target*. However, for all logical device configurations, only one initiator can be actively connected to the disk target. |

^a For library and autoloader configurations, you set iSCSI security for the library/autoloader only, not for the individual tape or RDX drives; the drives inherit the security setting from the library/autoloader.

Important As a best practice, set iSCSI security in the RDX QuikStation Remote Management Console first. Then, using the host's iSCSI initiator software, connect the host to your iSCSI targets on the RDX QuikStation.

> If you do not set security in the Remote Management Console first, any previously connected hosts will remain connected to the RDX QuikStation regardless of the security you set.

This section covers the following topics:

- Setting iSCSI Security in the Remote Management Console (page 34)
- Connecting iSCSI targets with CHAP for Windows (page 37)
- Connecting iSCSI targets with CHAP for Linux (page 40)
- Connecting iSCSI targets with CHAP for Macintosh (page 42)

SETTING ISCSI SECURITY IN THE REMOTE MANAGEMENT CONSOLE

To set iSCSI security:

- 1. Sign in to the RDX QuikStation Remote Management Console (see page 20).
- 2. Click the Logical Device tab, then select the iSCSI target from the tree view in the left pane.

| 🔍 iSCSI Devices | Properties | Connection |
|------------------------|-------------------|--|
| Disk Drive1 (Dock01) | Name | Value |
| Disk Drive3 (Dock03) | Vendor | TANDBERG |
| 🔜 Disk Drive4 (Dock04) | Product | RDX |
| Disk Drive5 (Dock05) | Revision | 0063 |
| Disk Drive6 (Dock06) | Serial Number | PD0000059008 |
| Disk Drive7 (Dock07) | iSCSI Name | iqn.2010-01.com.tandbergdata:storage.rdx1-1.0025 |
| | Target ID/LUN | 1/0 |
| | Status | Good |
| | Media Label | Disk1 |
| | Media State | Loaded |
| | | |
| | 4 | |

- **3.** Click the Security tab.
 - To set CHAP authentication, see page 35.
 - To set host authentication, see page 36.

Enabling CHAP Authentication

To add CHAP users (hosts) to an iSCSI target:

- 1. From the Security tab dialog window, make sure the CHAP Authentication option is shown in the pulldown window.
- 2. Click + Add.

| 4 🍭 iSCSI Devices | Properties 🥖 Connection | 🤴 Security |
|------------------------|---------------------------|------------|
| 🚔 Disk Drive1 (Dock01) | Tune: HHAD Authentication | |
| 📾 Disk Drive2 (Dock02) | | |
| 📾 Disk Drive3 (Dock03) | Allowed User | V |
| 📾 Disk Drive4 (Dock04) | | |
| 📾 Disk Drive5 (Dock05) | | |
| 📾 Disk Drive6 (Dock06) | | |
| 📾 Disk Drive7 (Dock07) | | |
| 📾 Disk Drive8 (Dock08) | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |
| | | |

The Add iSCSI User dialog box opens.

| Add iSCSI User | | × |
|--------------------|-----------|---|
| User Name: | | |
| Secret Key: | | |
| Verify Secret Key: | | |
| | | _ |
| | Add Close | |

- 3. In the User Name field, enter the host initiator name. By default, this is the host IQN name, which you can find from your iSCSI host initiator software application.
 - **Note:** You can create a unique user name in this field. If you do so, you must also change the default host initiator name in your host iSCSI initiator software to match this name.
- Type the CHAP secret. The secret must be 12 to 16 characters long. You can use any combination of letters, numbers, and the following special characters: -,!@#\$% ^ &*_.
- 5. Type the CHAP secret again to verify. Click Add.

Enabling Host Authentication

To add hosts that will be allowed to connect to an iSCSI target:

- 1. From the Security tab dialog window, select Host Authentication from the pulldown window.
- 2. Click + Add.



The Add iSCSI Host dialog box opens.

| Add iSCSI Host | | × |
|------------------|-----|---|
| Host IP Address: | | |
| | Add | |

- 3. Type the IP address for the network host.
- 4. Click Add.

CONNECTING ISCSI TARGETS WITH CHAP FOR WINDOWS

The following instructions describe how to connect your network host computers in Windows Server 2008 R2. Your iSCSI initiator interface may differ slightly depending on your Windows operating system.

To establish an iSCSI connection with CHAP enabled:

- 1. Open Microsoft iSCSI Initiator and click the Discovery tab.
- 2. Click Discover Portal...). (Your version of iSCSI initiator may use Add depending on your Windows OS.)

| nelis Discover | Y Favorite Targe | ets Volumes and Devices | RADIUS Configuration | |
|--|---|---|---------------------------|--|
| | . Haroneo rarge | | Conngaradio | |
| arget portals - | | | | |
| The system will look for Targets on following portals: | | | | |
| Address Port Adapter IP address | | | | |
| | | | · | |
| o add a targe | t portal, click Disco arget portal, select | over Portal. the address above and | Discover Portal | |
| To add a targe To remove a ta hen click Remo | t portal, click Disco arget portal, select ave. | over Portal. : the address above and | Discover Portal Remove | |
| Fo add a targe Fo remove a ta :hen click Remo | t portal, click Disco arget portal, select ove. | over Portal. : the address above and | Discover Portal | |

The Discover Target Portal dialog window opens.



- 3. In the Discover Target Portal dialog window, type the IP address for the RDX QuikStation. Click OK.
- 4. Click the Targets tab.

Depending on how you configured the logical device type for the QuikStation, you will see two to eight iSCSI targets. The screenshot below depicts a TandbergData StorageLibrary T24 configuration. The three iSCSi targets connect the library and two LTO-3 tape drives to the host.

| Junitiator Properties | |
|--|---|
| argets Discovery Favorite Targets Volumes and Devices | RADIUS Configuration |
| Quick Connect | |
| To discover and log on to a target using a basic connection, DNS name of the target and then click Quick Connect. | type the IP address or |
| Target: | Quick Connect |
| Discovered targets | |
| | Refresh |
| Name | Status |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 | Inactive |
| | Inactive |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 | Inactive |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 | Inactive |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 To connect using advanced options, select a target and ther click Connect. | Connect |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 To connect using advanced options, select a target and ther click Connect. To completely disconnect a target, select the target and then click Disconnect. | Connect Disconnect |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 To connect using advanced options, select a target and ther click Connect. To completely disconnect a target, select the target and then click Disconnect. For target properties, including configuration of sessions, select the target and click Properties. | Connect Disconnect Properties |
| iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 iqn.2010-01.com.tandbergdata:storage.MAGNUM-224 To connect using advanced options, select a target and ther click Connect. To completely disconnect a target, select the target and then click Disconnect. For target properties, including configuration of sessions, select the target and click Properties. For configuration of devices associated with a target, select the target and then click Devices. | Inactive Inactive Connect Disconnect Properties Devices |

- 5. For each iSCSI target, complete the following steps:
 - a. Select an iSCSI qualified name (IQN). Click Connect. (Your version of iSCSI initiator may use Log On...) depending on your Windows OS.)

The Connect to Target dialog window opens.

| Connect To Target | 5 W | 0 | R | D | × |
|--|------------------------------------|----------|-----|-------|----|
| Target name: | | | | | |
| 10-01.com.tandbergdata:storage.MAGNUM | 1-224-1.0025 | 5900898 | E6A | | |
| Add this connection to the list of Favorite This will make the system automatically a connection every time this computer rest | Targets. ttempt to res arts. | store th | e | | |
| 🔲 Enable multi-path | | | | | |
| Advanced. | C | Ж | | Cance | el |

b. Select the "Add this connection to the list of Favorite Targets" checkbox. Click Advanced...). The Advanced Settings dialog window opens.

| Advanced Settings | -? |
|---|--|
| General IPsec | |
| Connect using | |
| Local adapter: | Default |
| Initiator IP: | Default |
| Target portal IP: | Default 🗸 |
| | |
| CRC / Checksum | |
| Data digest | Header digest |
| 🗹 Enable CHAP log on | |
| CHAP Log on informati | on |
| CHAP helps ensure conn an initiator. | ection security by providing authentication between a target and |
| To use, specify the same | e name and CHAP secret that was configured on the target for this |
| initiator. The name will of specified. | default to the Initiator Name of the system unless another name is |
| | in 1001 05 eeu simee friedel is del |
| Name: | Ign.1991-05.com.microsorciisviab-isviab |
| Target secret: | •••••• |
| Perform mutual auth | entication |
| To use mutual CHAP, eit RADIUS. | her specify an initiator secret on the Configuration page or use |
| Use RADIUS to gene | rate user authentication credentials |
| Use RADIUS to authe | enticate target credentials |
| | |
| | OK Cancel Apply |

- c. Select the "Enable CHAP log on" checkbox.
- d. Type the target secret. This secret must match the CHAP secret that was set in the RDX QuikStation Remote Management Console for this particular iSCSI target.
- **Note:** The host initiator name, which is displayed above the target secret in the Advanced Settings dialog, must match the initiator user name that is specified in the Remote Management Console.
- e. Click OK, then click OK again in the Connect to Target dialog window.

The RDX QuikStation's iSCSI target should display as "connected" in the Discovered Targets dialog window, as shown below.

| gets Dis | covery | Favorite Targets | Volumes and Device: | s RADIUS | Configuration |
|--|---------------------------------|-------------------------------------|------------------------|---------------------|-----------------|
| uick Conn | ect | | | | |
| o discove | r and log | on to a target usin | ng a basic connection, | type the IP | address or |
| No name | | irget and then thick | QUICK CONNECC. | | |
| araati | | | | | uick Coppect |
| aracu | | | | | aion connicconn |
| | <u> </u> | | | | |
| iscovered | targets | | | | |
| iscovered | targets | | | | Refresh |
| iscovered Name | targets | | | Status | Refresh |
| iscovered Name iqn.2010- | targets 01.com.t | andbergdata:stora | age.MAGNUM-224 | Status | Refresh |
| scovered Vame qn.2010- qn.2010- | targets 01.com.t 01.com.t | andbergdata:stora:andbergdata:stora | age.MAGNUM-224 | Status Connected | Refresh 1 |

Important Before you shutdown or restart the RDX QuikStation for any reason, Tandberg Data recommends that you disconnect all iSCSI initiators.

CONNECTING ISCSI TARGETS WITH CHAP FOR LINUX

The following instructions describe how to connect iSCSI targets with CHAP authentication using Linux Open-iSCSI Initiator. Open-iSCSI Initiator is available for Redhat Enterprise Linux, CentOS, Fedora Linux, Debian, and Ubuntu operating systems. For more information about using Open-iSCSI with your specific Linux operating system, refer to your Linux distribution web site.

1. Install Open-iSCSI initiator.

To install the iSCSI initiator, enter the following command:

yum install iscsi-initiator-utils

For Ubuntu and Debian operating systems, enter the following command:

\$ sudo apt-get install open-iscsi

- 2. Configure CHAP authentication.
 - **a.** Open the **iscsid.conf** file using a vi text editor with the following command:

vi /etc/iscsi/iscsid.conf

b. Set the authmethod to CHAP by replacing *None* with *CHAP* so that the following line reads:

"node.session.auth.authmethod = CHAP"

c. Set the host initiator user name and CHAP password by uncommenting the following lines and replacing the variables with the actual host initiator name and password needed to connect to the iSCSI target.

#node.session.auth.username = (host initiator name)
#node.session.auth.password = (CHAP password)
#discovery.sendtargets.auth.username = (host
initiator name)
#discovery.sendtargets.auth.password = (CHAP
password)

d. Restart the iSCSI service by entering the following command:

/etc/init.d/iscsi start

3. Discover iSCSI Targets.

To discover the RDX QuikStation iSCSI targets, enter the following command, replacing the variable *RDX QuikStation IP address* with the actual IP address for your RDX QuikStation:

iscsiadm -m discovery -t sendtargets -p RDX QuikStation
ip address

Depending on how you configured the logical device type for the RDX QuikStation, you will see two to eight iSCSI targets.

4. Connect iSCSI targets.

To connect all discovered iSCSI targets, enter the following command:

```
# iscsiadm -m node -L all
```

To connect to an individual target, enter the following command, replacing *target IQN name* with the iSCSI qualified name (IQN) of the target and replacing *target portal* with the RDX QuikStation IP Address:

iscsiadm -m node -T target IQN name -p target portal --login

Once all iSCSI targets are connected, the initiator (Linux host) can use the iSCSI disks as a normal disk. It appears under/dev/sdx devices and you can format, mount iSCSI disks like a normal disk.

Important Before you shutdown or restart the RDX QuikStation for any reason, Tandberg Data recommends that you disconnect all iSCSI initiators.

CONNECTING ISCSI TARGETS WITH CHAP FOR MACINTOSH

The following instructions describe how to connect your Macintosh host computers with CHAP authentication using Atto Technology's Xtend SAN iSCSI initiator. Xtend SAN is compatible with Mac OS X 10.4.x and later. While we do not require you to use Xtend SAN as your third-party Mac iSCSI initiator, we recommend that you use an iSCSI initiator that has been tested or recommended for use with your chosen backup application.

To establish a secure iSCSI connection:

1. Open Xtend SAN iSCSI Initiator. Click the Discover Targets tab, then click Discover by DNS/IP.

| Hosts | | | Initiator | Discover Targets |
|------------------------------|-------------------------|------------------------------|-----------|--|
| 🖳 Tandberg-Data-ISVs-MacBook | Discover Configure t | Targets the static discov | very. | |
| | Address: | 10.1.50.211 | | |
| | Port: | 3260 | | |
| | CHAP | | | |
| | Target | User Name: | iqn | n.1995–12.com.attotech:xtendsan:ser.wq93 |
| - | Target | Secret: | | |
| | Mutual | Authentication: | | |
| | Initiato | r User Name: | | |
| | Initiato | r Secret: | | |
| | | | | |
| | | | | |
| | | | Finish | Cancel |

The following dialog window opens.

2. In the Address field, type the IP address for the RDX QuikStation. If you are using a port number other than the default, type the number in the Port field. Otherwise leave the default 3260 port number. Click Finish.

The dialog window displays the discovered iSCSI targets. Depending on how you configured the logical device type for the QuikStation, you will see two to eight iSCSI targets. The screenshot below depicts a configuration of eight removable disks.

| 0 0 | ATTO iSCSI Mac InitiatorXtend SAN |
|-------------------------------------|--|
| Hosts Tandberg-Data-ISVs-MacBook | ATTO iSCSI Mac InitiatorXtend SAN Initiator Discover Targets Discover by DNS/IP Discover by iSNS Discovered Targets iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590089e6a iqn.2010-01.com.tandbergdata:storage.rdx6-6.002590089e6a iqn.2010-01.com.tandbergdata:storage.rdx4-4.002590089e6a iqn.2010-01.com.tandbergdata:storage.rdx4-4.002590089e6a iqn.2010-01.com.tandbergdata:storage.rdx3-3.002590089e6a iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a |
| | Add Clear |

3. Highlight all the discovered targets and click Add.

The targets are displayed in the left pane with a red icon, signifying that they are not yet connected.

| 😑 🔿 🔿 🔿 ATTO iSCSI Mac Initia | torXtend SAN |
|---|---|
| Mattro iscSI Mac Initiat Hosts Indberg-Data-ISVs-MacBook-Pro.local iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590085 iqn.2010-01.com.tandbergdata:storage.rdx2-2.002590085 iqn.2010-01.com.tandbergdata:storage.rdx3-3.002590085 iqn.2010-01.com.tandbergdata:storage.rdx4-4.002590085 iqn.2010-01.com.tandbergdata:storage.rdx5-5.002590085 iqn.2010-01.com.tandbergdata:storage.rdx7-7.002590085 iqn.2010-01.com.tandbergdata:storage.rdx8-6.002590085 iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590085 iqn.2010-01.com.tandbergdata:storage.rdx8-8.002590085 | torXtend SAN Initiator Discover Targets Discover by DNS/IP Discover by iSNS Discovered Targets iqn.2010-01.com.tandbergdata:storage.rdx8-8.0025 iqn.2010-01.com.tandbergdata:storage.rdx7-7.0025 iqn.2010-01.com.tandbergdata:storage.rdx5-5.0025 iqn.2010-01.com.tandbergdata:storage.rdx4-4.0025 iqn.2010-01.com.tandbergdata:storage.rdx3-3.0025 iqn.2010-01.com.tandbergdata:storage.rdx2-2.0025 iqn.2010-01.com.tandbergdata:storage.rdx1-1.0025 |
| | Add Clear |

- 4. For each iSCSI target, complete the following steps to connect the target:
 - a. Select an iSCSI qualified name (IQN) from the left pane.

The target's port information opens in the center pane.

| Setup Status LUNs | | | | | | |
|---|--|--|--|--|--|--|
| ame: iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a | | | | | | |
| lias: | | | | | | |
| Ports | | | | | | |
| Visible | Auto Login | | | | | |
| | | | | | | |
| | Setup Status LUNs .tandbergdata:storage.rd Visible | | | | | |

- **b.** Select the "Visible" checkbox and the "Auto Login" checkbox.
- c. Click Security.

The Configure Security dialog window opens.

| Target User Name: | iqn.1995-12.com.atto | tech:xtendsan:ser.wq939 | 4wa66d |
|------------------------|----------------------|-------------------------|--------|
| Target Secret: | | | |
| Mutual Authentication: | | | |
| Initiator User Name: | | | |
| Initiator Secret: | | | |

- d. Type the target secret. This secret must match the CHAP secret that was set in the RDX QuikStation Remote Management Console for this particular iSCSI target.
- **Note:** The target user name (the host's IQN name), must match the user name that is specified in the iSCSI Management settings of the Remote Management Console.
- e. Click Ok.
- f. Click the Status tab.

The target's connection status is displayed.

| (| Setup Status LUN | s | | | | |
|---|------------------|------------|--|--|--|--|
| ame: iqn.2010-01.com.tandbergdata:storage.rdx1-1.002590089e6a | | | | | | |
| Alias: | | | | | | |
| Ports | | | | | | |
| Network Node | Status | Auto Login | | | | |
| 10.1.50.211:3260 | Not Connected | Yes | | | | |

g. Select the Network Node and click Login.

The iSCSI targets will display with a green icon when they are connected to the RDX QuikStation, as shown below.

| 0 O ATTC | iSCSI Mac InitiatorXten | d SAN | |
|---|---|---------------------|----------------------------------|
| Hosts Tandberg-Data-ISVs-MacBook-Pro.local iqn.2010-01.com.tandbergdata:storage.rdx1-1. iqn.2010-01.com.tandbergdata:storage.rdx2-2. iqn.2010-01.com.tandbergdata:storage.rdx3-3. iqn.2010-01.com.tandbergdata:storage.rdx4-4. | Name: iqn.2010-01.co Alias: _ Ports | Setup Status | LUNs rage.rdx3-3.002590089e6a |
| iqn.2010-01.com.tandbergdata:storage.rdx5-5. iqn.2010-01.com.tandbergdata:storage.rdx6-6. iqn.2010-01.com.tandbergdata:storage.rdx7-7. iqn.2010-01.com.tandbergdata:storage.rdx8-8. | Network Node 10.1.50.211:3260 | Status Connected | Auto Login Yes |

Before you shutdown or restart the RDX QuikStation for any reason, Tandberg Data recommends that you disconnect all iSCSI initiators.

4

SYSTEM CONFIGURATION

This chapter describes the RDX QuikStation system settings that a user with administrator privileges can view and modify.

The chapter includes the following sections:

| System settings overview page 48 |
|--|
| Setting date and time page 49 |
| Configuring network information page 50 |
| Setting email notifications page 53 |
| Setting RDX QuikStation options page 54 |
| Managing user groups page 60 |
| Recovering the built-in administrator password page 64 |

SYSTEM SETTINGS OVERVIEW

As a user with administrator permissions, the following functions are available for you to view and modify from the Remote Management Console's System Settings menu.

| System Setting Menu Name | Description | |
|-----------------------------|---|--|
| Date and Time | Allows you to set the date and time for the RDX QuikStation (see page 49). | |
| Network Information | Allows you to modify the RDX QuikStation name, assign a static IP address, configure DNS server addressing information, and set a port bonding mode (see page 50). | |
| Notification | Allows you to specify email addresses where messages for error reporting and password recovery are sent, enable SNMP authentication, and customize email sender and subject line text (see page 53). | |
| User Management | Allows an administrator to add, remove, and edit users that are allowed access to the RDX QuikStation Remote Management Console (see page 60). | |
| Convert Device Type | Allows you to change the logical device type for the RDX QuikStation. For more information about logical device types, see | |
| | Note: If you change from a virtual library configuration to a removable disk configuration, you will need to reformat the RDX cartridges (see page 86). | |
| | sure there are no active host connections to the iSCSI targets before doing so. | |
| Options | The Options dialog window is divided into four tabs: System, Removable Disk, Automation, and Diagnostics. See "Setting RDX QuikStation Options" on page 54 for more information. | |

 Table 4-1
 RDX QuikStation System Settings

SETTING DATE AND TIME

Note: You must have administrator privileges to set the date and time.

To set the date the time and time for your RDX QuikStation:

 From the Remote Management Console main menu, select System Settings > Date/Time.



The Date/Time Settings dialog window opens.

| Date/Time Settings | | × |
|--------------------|--|---|
| Time Zone: | America/Denver | |
| System Date: | 09/16/2011 | |
| System Time: | 11 🗘 : 59 🗘 | |
| | nized with an internet time server (NTP) — | |
| | | |
| | Update Close | |

- 2. Choose one of the following options:
 - Modify the time zone, system date, and system time fields. Click Update.
 - Select the "Keep synchronized with an internet time server (NTP)" checkbox. Type the NTP server IPv4 address or fully qualified domain name in the Time Server field. Select your time zone. Click Update.
 - **Note:** At least one city from every time zone region is listed. If you do not see your city listed in the pull-down menu, look for another city that is in the same time zone as your city. For example, "America/Los Angeles" represents the Pacific time zone.

CONFIGURING NETWORK INFORMATION

Note: You must have administrator privileges to configure network information.

To configure network information for the RDX QuikStation:

 From the Remote Management Console main menu, select System Settings > Network.

| | 🍪 S | System Settings 🔹 🎤 Sy | vstem Maintenance 🔻 🛛 🕘 Refresh 🛛 📃 Sign |
|---|-----|------------------------|--|
| | | Date/Time | |
| 1 | + | Network fm | |
| | | Notification | |
| • | æ | User Management | |
| | 2 | Convert Device Type | |
| - | × | Options | |
| | | | |

The Network Settings dialog window opens.

| Network Settings | | | | |
|-------------------------|---------------|--------------------|--|--|
| Basic Advance | asic Advanced | | | |
| Host Name: | QS989 | 0 | | |
| Search Domain: | domain | mycompany.com | | |
| 🔘 Obtain IP addr | ess autom | natically (DHCP) | | |
| ⊙ Use the follow | ing IP add | dress | | |
| | | | | |
| IP Address: | | 10.1.50.211 | | |
| Subnet Mask: | | 255.255.255.0 | | |
| Default Gateway: | | 10.1.50.254 | | |
| Configure DNS server | | addresses manually | | |
| Primary DNS: | | 10.1.10.1 | | |
| Secondary DNS: | | 10.1.10.2 | | |
| | | | | |
| Change (Im) Reset Close | | | | |

- 2. Modify the host name if desired. The default RDX QuikStation host name uses "QS" and the last four digits of the unit's serial number.
- **3.** Add or modify the search domain (DNS lookup domain). This field is optional.
- 4. Choose to either obtain network addressing automatically via DHCP or set static addressing information. Tandberg Data recommends you set a static IP address to prevent inadvertent address changes.
 - **Note:** If you are setting static network information, the default gateway is an optional field.
- 5. Enter a primary and secondary DNS server address. These fields are optional.

- 6. Select an ethernet port configuration. By default, the port setting is set for Active Backup, or Failover. In most cases, this default setting will suffice for your network environment. See Configuring Advanced Network Settings in the section below for more information.
- 7. Click Change.

CONFIGURING ADVANCED NETWORK SETTINGS

The Advanced tab section of the Network Settings dialog window displays MAC address and link speed information and also provides port configuration options (see Table 4-2 on page 52).

| Network Settings | × |
|---|---|
| Basic Advanced | |
| MAC Address(0); 00:25:90:08:9e:6b | |
| MAC Address(1): 00:25:90:08:9e:6a | |
| Link Speed: 1Gbps | |
| Port Configuration | |
| Note: Refer to the product manual for a detailed description of available options. | |
| Active Backup (Failover) | |
| | |
| | |
| | |
| | |
| | |
| | |
| Change Reset Close | |

To change the port configuration:

- 1. Select the port configuration from the pulldown menu.
- 2. Click Change.

Port Configuration Options

The RDX QuikStation has two Gigabit Ethernet ports to provide failover protection and high availability network connectivity. In network environments, it is a common practice to configure two or more external ports as a single, logical, bonded port using an industry standard feature called Link Aggregation, also known as port bonding or port trunking.

The behavior of the logically-bonded ports depends upon the mode. For the RDX QuikStation, you can configure port bonding for either active backup/failover or adaptive load balancing.

The following table describes the two port bonding modes that are available in the RDX QuikStation.

| Port bonding mode | Description | |
|--|---|--|
| Active-backup (failover) | The Active-backup or failover mode is the default configuration for the RDX QuikStation. This mode uses one port (adapter) and will only switch to the other port if the active port fails. | |
| Adaptive load balancing (balance-alb) | Adaptive Load Balancing provides load balancing of transmit and receive traffic. It also provides automatic failover if one cable fails. This option works with any switch and is best used in a multi-disk environment that has multiple iSCSI host connections. You might consider using this option if you have configured your RDX QuikStation as eight RDX drives, and you are using most or all of the eight iSCSI disk targets. | |

Table 4-2Port bonding options

SETTING EMAIL NOTIFICATIONS

Note: You must have administrator privileges to set email notifications.

The Notifications Settings dialog window consists of three tabs: basic, authentication, and customization. The authentication tab allows you to add a user name and password for SMTP authentication, if your email server requires it. The Customization tab allows you to define the "From" email address and the subject text for email messages that are sent by the RDX QuikStation.

SETTING BASIC EMAIL NOTIFICATION

 From the Remote Management Console main menu, select System Settings > Notification.



- 2. Type one or two valid email addresses in the email recipient fields. The RDX QuikStation will send any system error messages, such as a fan failure or an RDX dock failure, to the addresses that you specify. The RDX QuikStation will also send any password recovery requests to these addresses.
- **3.** Enter a fully-qualified email server domain name. For example, *mailhost.mycompany.com*. You can also enter a valid IPv4 address. For example, *192.168.1.101*.
- 4. If your email server uses SSL/TSL, select the checkbox to enable it.
- 5. Click Change.

SETTING SMTP AUTHENTICATION

- 1. Click the Authentication tab from the Notification Settings dialog window.
- 2. Type the SMTP user name and password for your email server.
- 3. Select the authentication method from the pulldown menu either "plain" or "CRAM-MD5."
- 4. Click Change.

CUSTOMIZING EMAIL SUBJECT AND "FROM" ADDRESS

- 1. Click the Customization tab from the Notification Settings dialog window.
- 2. Type your desired "From" email address using standard email format, such as "myQuikStation@mycompany.com." The "From" email text does not need to be a valid email address.
- 3. If desired, type a custom subject text. This text will be used in all emails that are generated by your RDX QuikStation.
- 4. Click Change.

SETTING RDX QUIKSTATION OPTIONS

The Options dialog window is divided into four tabs: System, Removable Disk, Automation, and Diagnostics.

Refer to the table below for information about each tab.

 Table 4-3
 RDX QuikStation option settings

| Tab | Option | Description |
|----------------|---|--|
| System | Automatic Firmware Updates | Allows you to turn off the option to automatically check for firmware updates when you log into the Remote Management Console. |
| | LCD Language | Allows you customize the language that is displayed in the RDX QuikStation's LCD panels. The language options are English, French, German, or Spanish. |
| Removable Disk | Unsafe Eject | Allows you to enable the "Unsafe Eject" mode for RDX cartridges that have an established host connection (see page 56). |
| Automation | Auto-insert (import/export functionality) | Allows you to control how the RDX QuikStation handles tape library import/export slot functionality (see page 57). |
| Diagnostics | Log Level | Allows you to change the log level from Normal to Detail or Extend. Tandberg Data recommends you only change the log level if you are instructed to do so by Tandberg Data Technical Support. |
| | | Note: If you change the log level, the system will default to "normal" once the Management Console browser session is closed. |
| | Remote Access | Allows you to control remote access to the RDX QuikStation via SSH (Secure Shell). The default setting disables SSH remote access. |

To access the Options dialog window:

From the Remote Management Console main menu, select System Settings > Options.



The Options dialog window opens with the System tab selected.

| Options | × |
|--|---|
| System Removable Disk Automation Diagnostics | |
| Firmware Update Check Option | |
| LCD Language | |
| Change LCD panel language setting. English | |
| | |
| | |
| | |
| OK Cancel | |

SETTING REMOVABLE DISK OPTIONS

Note: The removable disk option only applies to disk configurations.

You can enable the "Unsafe Eject" mode for RDX cartridges that have an established host connection.

By default, when an RDX dock is configured as a logical disk drive that is connected to a host computer, you cannot manually eject the cartridge using the RDX dock eject button or from using the eject function in the Remote Management Console. This is because the connected host computer's operating system has reserved the cartridge.

Enabling the unsafe eject option allows you to override the default and manually eject an RDX cartridge. This option is not recommended because you could potentially lose data on the RDX cartridge if you use unsafe eject. We recommend that you eject the cartridge from your backup software application or from the host computer's eject function.

To enable Unsafe Eject:

1. Click the Removable Disk tab from the Options dialog window.



- 2. Select the "Enable Unsafe Eject" checkbox.
- 3. Click OK.

SETTING THE AUTO-INSERT OPTION

The auto-insert option controls how the RDX QuikStation handles tape library import/export slot functionality.

The following table describes RDX QuikStation behavior when this option is turned off and when it's turned on.

| Auto-insert option setting | Behavior | |
|-------------------------------|--|--|
| OFF | When a single RDX cartridge is inserted into any dock, it will be mounted into the virtual IE port, awaiting an import command from your backup software. When multiple cartridges are inserted, the first cartridge that is detected by the library will be mounted to the virtual IE port. The remaining inserted cartridge will not display in the Remote Management Console until you import them one at a time via your backup software application (see Figure 4-1 and Figure 4-2). | |
| | Note: Refer to your backup software documentation for tape library import/export commands and recommended use. | |
| ON (Default) | When one or more RDX cartridges are inserted, the RDX QuikStation automatically moves them into an available logic slot (see Figure 4-3 and Figure 4-4). | |
| | The auto-insert option eliminates the need to use your backup software to import each inserted cartridge. If the slot corresponding to the physical dock is available, that slot will be used by the RDX QuikStation. | |
| | However, you can still perform an export from the backup application to eject the media from a slot and physical dock. | |

 Table 4-4
 Auto-insert (I/E port behavior) option settings

To change the auto-insert option:

1. Click the Automation tab from the Options dialog window.



2. Select your desired setting. Click OK.

Note: The default is the "On" setting. See Table 4-4.

| Physical Device | 🛃 Logical Device | |
|--|---------------------------------|-----------------------------------|
| 🛃 Convert Device Type | | 🔤 Move Media 🛛 🎇 Advanced Options |
| SCSI Devices StorageLibrary T40 Tape Drive1 Tape Drive2 | Properties Name Flement Address | Value |
| Tape Drive2 Tape Drive3 Tape Drive4 Tape Slots Slot1 Slot2 Slot3 Slot4 Slot5 Slot5 Slot6 | Media Label Media Usage | 601391L3 0.63/1606B (0.4%) |
| I Slot8 | | |

Figure 4-1 A logical view of a tape library with the auto-insert option turned OFF. Note that only the IE Port displays as populated until you import the other cartridges one at a time with your backup software.

| Physical Device | 😂 Logical Device | |
|--|-------------------------|-------------------------------------|
| | | 📤 Eject । 🧷 Erase 🚽 Clone 👛 Tes |
| 4 🥝 QS9890 | Cock 06 Properties | |
| Dock01 (7A007AL3) | Name | Value |
| Dock02 (127BR9L3) | Dock Status | Good |
| Dock04 (7A00EAL3) | Dock LCD Text | 601391L3 |
| 🖿 Dock05 (B25CSWL3) | Dock FW Rev | 0063 |
| 🛤 Dock06 (601391L3) | Cartridge Status | Good |
| ■ Dock07 (766906L3) ■ Dock08 (588189L3) | Cartridge Serial Number | 859900601391 |
| | Cartridge Label | 601391L3 |
| | Cartridge Capacity | 160GB |
| | Cartridge Temp(C) | 22C degrees |
| | Write Protected | Not protected |
| | | |
| | | |
| | | |
| | | |
| | | |

Figure 4-2 A physical view of a tape library with the auto-insert option turned OFF. Note that four cartridges are displayed in their physical slots, where in Figure 4-1, only one cartridge is displayed in the IE port.

| Physical Device | logical Device | |
|--------------------------------|-----------------|----------------------------------|
| 🛃 Convert Device Type | | 🔤 Move Media 🎇 Advanced Option |
| 🗸 🍓 iSCSI Devices | Properties | |
| StorageLibrary T40 Tape Drive1 | Name | Value |
| Tape Drive2 | Element Address | 1 |
| Tape Drive3 | Media Label | 7A007AL3 |
| Tape Drive4 | Media Usage | 0/750GB (0%) |
| IE Port A Pape Slots | | |
| 🔤 Slot1 (Dock01) | | |
| 🔤 Slot2 (Dock02) | | |
| 🔤 Slot3 (Dock03) | | |
| 🔤 Slot4 (Dock04) | | |
| 🔤 Slot5 (Dock05) | | |
| 🔤 Slot6 (Dock06) | | |
| 🔤 Slot7 (Dock07) | | |
| 🔤 Slot8 (Dock08) | | |

Figure 4-3 A logical view of a tape library with the auto-insert option turned ON. Note that cartridges automatically mount to logical slots that correspond with their physical dock locations.

| Physical Device | Jugical Device | |
|---------------------|-------------------------|-------------------------------------|
| | | 📤 Eject 🧷 Erase 🚽 Clone 📫 Tes |
| 4 | Cock 01 Properties | |
| Dock01 (7A007AL3) | Name | Value |
| Dock02 (127BR9L3) | Dock Status | Good |
| Dock04 (7A00EAL3) | Dock LCD Text | 7A007AL3 |
| Dock05 (B25CSWL3) | Dock FW Rev | 0063 |
| Dock06 (601391L3) | Cartridge Status | Good |
| Dock07 (766906L3) | Cartridge Serial Number | 575466F0 |
| - Dockoo (30010923) | Cartridge Label | 7A007AL3 |
| | Cartridge Capacity | 750GB |
| | Cartridge Temp(C) | 23C degrees |
| | Write Protected | Not protected |
| | | |
| | | |
| | | |
| | | |

Figure 4-4 A physical view of a tape library with the auto-insert option turned ON. Note that the cartridges' physical locations correspond with their logical tape slot locations (shown in Figure 4-3).

MANAGING USER GROUPS

Note: You must have administrator privileges to manage user groups.

The RDX QuikStation allows you to set up three types of groups: administrator, manager, and user.

This section contains the following topics:

- Group permissions overview (page 61)
- Adding users to a group (page 61)
- Removing users from a group (page 62)
- Editing user information (page 63)

GROUP PERMISSIONS OVERVIEW

The following table describes the functionality each group can access in the Remote Management Console.

 Table 4-5
 RDX QuikStation Group Permissions

| User | Manager | Administrator |
|--|---|---|
| Users have permission to do the following: | Managers have permission to do the following: | Administrators have access to all functionality. |
| change their log-in password. view physical and logical device information. | view group information. view iSCSI Management security and connection status. | This includes permission to:set system and network configuration. |
| refresh/update device information manually. | set options for automatic web page refresh and diagnostic log level. change their log-in password. view physical and logical device information. refresh/update device information manually. | set iSCSI security. add, remove, and edit groups. format/erase RDX cartridges. clone data on RDX cartridges. convert logical device type. update firmware. perform system diagnostic tests. access all other system maintenance tasks. |

ADDING USERS TO A GROUP

To add users to a group:

1. From the Remote Management Console main menu, select System Settings > User.



The User Management dialog window opens.

| er Managem | | | |
|------------|-------------------------|--------------------------------|--|
| Add 🔾 💭 R | emove 🛛 🌪 Edit Password | | |
| Iser") | Group | Description | |
| admin | Administrator | built-in administrator account | |
| user | User | built-in user account | |
| | | | |
| | | | |
| | | | |
| | | | |
| | | | |
| | | | |
| | | | |

- 2. Click +Add
- **3.** Type a user name and password in the applicable fields. The following conditions apply:
 - User names are not case sensitive and can only contain letters, numbers, and the "_" character. There is no limit on the number of characters.
 - Passwords are case sensitive and must be 6 to 30 characters long. Passwords can contain any combination of letters, numbers, and the following special characters: -,!@#\$% ^ &*_.
- 4. Select the group you want to assign the user to from the pulldown menu. For a description of the permissions available to each group, see page 61.
- 5. Type a description to identify the user, if desired.
- 6. Click Add.

REMOVING USERS FROM A GROUP

To remove users from a group:

1. From the User Management dialog window (go to System Settings > User), select the user that you want to remove. Click <u>-Remove</u>.

| User Management | | | × | |
|------------------------------|---------------|--------------------------------|-------|--|
| O Add Remove % Edit Password | | | | |
| User 🖑 | Group | Description | | |
| admin | Administrator | built-in administrator account | | |
| user | User | built-in user account | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | Close | |

- 2. Type your RDX QuikStation administrator password. Click OK.
- 3. Click Yes to confirm.
EDITING USER INFORMATION

You can edit a user name, the group assignment, description, and password.

Note: For the built-in administrator account, you can only edit the password.

Editing the user name, group, or description

1. From the User Management dialog window (go to System Settings > User), double-click in the field you want to edit.

The field becomes editable, as shown below.

| User Managem | ent | | | × |
|--------------|--------------------|--------|--------------------------------|-------|
| 🕃 Add 🛛 🤤 R | emove 🛛 🌪 Edit Pas | ssword | | |
| User | Group | | Description | |
| admin | Administrator | r | built-in administrator account | |
| user | User | _lh_ 💙 | built-in user account | |
| | | U | | |
| | | | | Close |

- 2. Make your desired changes. Click Enter.
- **Note:** User names can only contain letters, numbers, and the "_" character. There is no limit on the number of characters.

Editing a password

- 1. From the User Management dialog window (go to System Settings > User), select the user.
- 2. Click Edit Paswword
- 3. Type a new password. Type the password again to verify. Click OK.
- **Notes:** For users assigned to administrator and manager groups, you must first enter your RDX QuikStation administrator password before editing the password.

Passwords are case sensitive and must be 6 to 30 characters long. Passwords can contain any combination of letters, numbers, and the following special characters: -,!@# $\%^{\&}_{*}$.

RECOVERING THE BUILT-IN ADMINISTRATOR PASSWORD

If you have forgotten the built-in administrator password, you can reset the password via email or via USB.

Note: The built-in administrator user name is **Admin** (cannot be modified) and the default password is **Admin!** (case sensitive).

If you have forgotten the user name and password for any other account, contact the RDX QuikStation administrator, or sign in as the built-in administrator account to edit the password (see page 63).

RESETTING THE PASSWORD VIA EMAIL

1. From the RDX QuikStation Sign-in dialog window, click the ? in the upper-right corner.

| Sign In to the R | DX QuikStation | |
|-------------------------|----------------|-----------------------|
| User Name: Password: | User Name | Forgot your password? |
| | | |
| | Sign I | n |

2. Select the Administrator radio button. Click Continue

The Administrator Password Reset dialog window opens.



3. Select Request Key).

The RDX QuikStation sends an email message to the address that was specified in the Remote Management Console (see page 53). The request key will remain valid for one hour.



- 4. Click OK and leave the Administrator Password Reset dialog window open.
 - **Note:** If you close the browser session, the request key that was sent via email becomes invalid.
- 5. Copy the request key from the email message and paste it into the Reset Key field.

| Administrator Password Reset | | | | | | | |
|------------------------------|--------------|--|--|--|--|--|--|
| Reset Key: | | | | | | | |
| New Password: | | | | | | | |
| Verify New Password: | | | | | | | |
| | | | | | | | |
| | Reset Cancel | | | | | | |

- 6. Type a new password. Type the password again to verify. Click Reset).
 - **Note:** Passwords are case sensitive and must be 6 to 30 characters long. Passwords can contain any combination of letters, numbers, and the following special characters: $-,!@#\%^&*_.$

RESETTING THE PASSWORD VIA USB

If an email notification address is not specified in the Remote Management Console, you can reset the built-in administrator password via USB by creating a **password.reset** file.

Important The Tandberg Data-supplied USB flash drive includes the file **network.conf**. Make sure that this file name is changed to another name, for example **network.conf.used**, before you insert the flash drive into the RDX QuikStation.

Renaming the file prevents any inadvertent network configuration changes to your RDX QuikStation.

To create the **password.reset** file and reset the password via USB:

- 1. Insert the USB flash drive that was included in your RDX QuikStation accessory kit box into your computer. If you cannot find the flash drive, you can use any USB flash drive that is formatted with a VFAT file system.
- 2. Navigate to the flash drive and create a blank text file on the drive. (In Windows, right-click in the directory window, and select New>Text Document, as shown below.)

| 😂 KINGST | ON (E:) | | | |
|-----------|-----------------|--|--------|---|
| File Edit | View Favorites | Tools Help | | |
| G Back | • 🕥 - 🏂 | Search 🔀 Folders | • | |
| Address 🥯 | E:\ | | | 💌 🄁 Go |
| Folders × | Name 🔺 | Siz | е Туре | Date Modified |
| | network.conf.us | ed View Arrange Icons By Refresh Paste Paste Shortcut Undo Copy | Ctrl+Z | 10/25/2010 12:30 PM |
| | | New Properties | | Eolder Shortcut Flash ActionScript File Briefcase Microsoft Word Document Adobe FrameMaker Document Microsoft PowerPoint Presentation Adobe Photoshop Image 12 Paint Shop Pro 7 Image Text Document Microsoft Vicio Proving |

3. Rename the text file to **password.reset**. (In Windows, click Yes) to confirm the file name change.)

| Folders × | Name 🔺 | Size | Туре | Date Modified | |
|-------------------|--------------------------------------|---|--|--|--|
| 🕝 Desk 🔺 🗆 🔒 N | network.conf.used password.reset | 3 KB 0 KB | USED File Text Document | 10/25/2010 12:30 PM 2/25/2011 2:20 PM | |
| | | | | | |
| ∣ 🛃 ⊟ ⇒ ⊕ | Rename | | | \mathbf{X} | |
| | 1 | If you change a file nam Are you sure you want (| e extension, the file may to change it? | / become unusable. | |
| H H H | | <u>Y</u> es | <u>N</u> o | | |
| ∃ ⊞ TVUVU | | | | | |
| 🛨 🔽 | | | | | |

4. Remove the USB flash drive from your computer and insert it into the RDX QuikStation's USB slot located on the center-front panel.

The USB symbol on the center-front panel flashes and the LCD dock panels display "RstPasswd."

- 5. Remove the USB flash drive from the RDX QuikStation when the USB symbol stops blinking and the LCD dock panels flash "Done."
- 6. Connect to the RDX QuikStation Remote Management Console (see page 20). From the Sign-in dialog window, click the ? in the upper-right corner.

| Sign In to the R | DX QuikStation | Sp. | |
|------------------|----------------|---------|-----------------------|
| User Name: | User Name | | Forgot your password? |
| Password: | - | | |
| | | 1.6 | |
| | | | |
| | | | |
| | | Sign In | |

7. Select the Administrator radio button. Click Continue.

The Administrator Password Reset dialog window opens.

| Click the Request Key button to send a password reset key to the email address specified in the Remote Management Console. Request Key Reset Password 2. Reset Administrator password via USB Insert a USB memory stick loaded with the password reset file into th USB slot on the front of the RDX QuikStation. When the USB symbo | Reset Administrator Password via email | |
|--|---|-----------------------|
| Request Key Reset Password 2. Reset Administrator password via USB Insert a USB memory stick loaded with the password reset file into th USB slot on the front of the RDX QuikStation. When the USB symbo | lick the Request Key button to send a password reset key to th smail address specified in the Remote Management Console. | e |
| Reset Password Reset Administrator password via USB Insert a USB memory stick loaded with the password reset file into th USB slot on the front of the RDX QuikStation. When the USB symbo | Request Key | |
| 2. Reset Administrator password via USB Insert a USB memory stick loaded with the password reset file into th USB slot on the front of the RDX QuikStation. When the USB symbo | Reset Password | |
| Insert a USB memory stick loaded with the password reset file into th USB slot on the front of the RDX QuikStation. When the USB symbo | Reset Administrator password via USB | |
| LED on the front of the {product} stops blinking, click the Reset Password button below. Refer to the product manual for information about the password reset file. | nsert a USB memory stick loaded with the password reset file int JSB slot on the front of the RDX QuikStation. When the USB syn LED on the front of the {product} stops blinking, click the Reset Password button below. Refer to the product manual for informa about the password reset file. | o the hbol tion |

- 8. Click Reset Password from the "Reset Administrator Password via USB" section of the dialog window.
- 9. Type a new password. Type the password again to verify. Click Reset. Click OK.
 - **Note:** Passwords are case sensitive and must be 6 to 30 characters long. Passwords can contain any combination of letters, numbers, and the following special characters: $-,!@#\$\%^\&*_.$

5

LOGICAL DEVICE MANAGEMENT

This chapter describes the Remote Management Console's Logical Device interface.

The chapter includes the following sections:

| Logical device type overview page 70 |
|---|
| Choosing a logical device type page 71 |
| iSCSI devices overview page 72 |
| Managing disks and tape drive configurations page 74 |
| Managing tape library and autoloader configurations page 76 |

LOGICAL DEVICE TYPE OVERVIEW

The following table compares the logical device types that the RDX QuikStation can emulate.

 Table 5-1
 Comparison of logical device types

| Logical Device Type | Allowed active iSCSI connections per target | Number of iSCSI targets | Reported inquiry string | Available I/E slot | Available virtual tape drives | iSCSI security | |
|--|--|----------------------------|----------------------------|--------------------------------|-------------------------------------|---|--|
| RDX drives ^a | | Eight | RDX | n/a | n/a | You can set two types of | |
| Tape drives | One active | Eight | Ultrium 3-SCSI | n/a | Eight | authentication and host authentication. Up to 10 initiator users or hosts are allowed for the drive target. | |
| Disk autoloader | host connection | Тwo | RDX DiskLibrary | yes (see | n/a | You can set two types of iSCSI security: CHAP | |
| T24 library | is allowed for each | Three | Magnum 224 | page 57) | Two | authentication and host authentication. Up to | |
| T40+library | iSCSI target. | Five | StorageLoader | | Four | 10 initiator users or | |
| StorageLoader | | Two | StorageLoader | | One | hosts are allowed for the library target. You cannot set security on the drive target; the | |
| Tape autoloader | | Two | StorageLoader | | One | | |
| Hybrid –T24 Library and RDX drives | | Seven | Magnum 224 and RDX | yes for the tape library | Two | drive inherits security from the library or autoloader. | |
| Hybrid – StorageLoader LTO and RDX drives | | Six | StorageLoader and RDX | n/a for the RDX drives | One | | |

^a The RDX drive is the recommended device type for disk configurations. The removable disk device type is available for legacy users.

Note: If you select the RDX Drive configuration, you must also install the RDX Utility, which is available from a link in the Logical Device Type Settings dialog window.

CHOOSING A LOGICAL DEVICE TYPE

There are many choices to look at when deciding which logical device type is best for your data storage needs. Some questions to consider are:

- Does your back-up application support removable disk or does it support tape only?
- Is your size of your backup job greater or less than the size of the physical RDX cartridge? For example, if you have 1 TB RDX cartridges, will your backup data fit on this one cartridge, or will you need to span the data across two or more cartridges?
- Do you require multiple, simultaneous backup jobs? For example, you may need to schedule multiple, simultaneous backup jobs to meet your backup window requirement. Simultaneous backups also allow you to break up your job to fit within the size of a single RDX cartridge.
- What backup application are you using? Does your backup application support spanning across multiple removable disk targets? (Applications that do support spanning for removable disk include Retrospect, ArcServe, and CommVault.)
- Does your backup application include a tape library unlimited drive option, which requires an extra license fee? With tape backups, if you require multiple, simultaneous backup jobs, you typically need to purchase a tape library option. If your backup application does not include a tape library option, you can choose the RDX drive logical device configuration, which does not require an extra license fee to perform multiple, simultaneous backups.

| Backup application support removable disk? | Backup job greater than size of RDX cartridge? | Need multiple, simultaneous backups? | Backup application support spanning for removable disk? | Unlimited tape drive option? (requires extra license fee) | Choose this logical device type: |
|---|--|--|---|--|---|
| Yes | No | Either Yes or No | No | No | Eight RDX Drives |
| Yes | Yes ^a | | Yes | No | |
| No | No | Either Yes or No | No | No | Eight LTO-3 Tape Drives |
| No | Yes | No | No | Yes | StorageLoader LTO |
| No | Yes | Yes | No | Yes | T24 StorageLibrary with two LT0-3 tape drives |
| No | Yes | Yes | No | Yes | T40+ StorageLibary with four LTO-3 tape drives |

Table 5-2 Choosing a logical device type

^a If yes, your backup application must support spanning for removable disk targets.

ISCSI DEVICE OVERVIEW

You can view general information about a logical device's iSCSI targets from the iSCSI Devices view in the Remote Management Console.

To navigate to the iSCSI Devices view:

• From the Logical Device interface, select iSCSI Devices from the tree view in the left pane.

| Physical Device | 😹 Logical Device | | |
|--|------------------|-----------------|---|
| 🛃 Convert Device Type | | | 🔤 Move Media 🛛 🎇 Advanced Options |
| iSCSI Devices رام | iscsi Devices | | |
| Disk Drive1 (Docko1) | Product Name | Serial Number 🔺 | iSCSI Name |
| Disk Drive3 (Dock03) | RDX | PD000000007 | iqn.2010-01.com.tandbergdata:storage.rdx2-2 |
| Disk Drive4 (Dock04) Disk Drive5 (Dock05) Disk Drive6 (Dock06) | RDX | PD0000059008 | iqn.2010-01.com.tandbergdata:storage.rdx1-1 |
| | RDX | PD0001752904 | iqn.2010-01.com.tandbergdata:storage.rdx5-5 |
| | RDX | PD0016503602 | iqn.2010-01.com.tandbergdata:storage.rdx7-7 |
| Disk Drive? (Dock07) | RDX | PD0026582303 | iqn.2010-01.com.tandbergdata:storage.rdx6-6 |
| Disk Drive8 (Docku8) | RDX | PD0037327805 | iqn.2010-01.com.tandbergdata:storage.rdx4-4 |
| | RDX | PD0050598506 | iqn.2010-01.com.tandbergdata:storage.rdx3-3 |
| | RDX | PD0056830001 | iqn.2010-01.com.tandbergdata:storage.rdx8-8 |
| | < | III | <u> </u> |

Figure 5-1 iSCSI Devices view

By default, the iSCSI Devices window displays the product name, serial number and iSCSI name for the logical device's iSCSI targets. Figure 5-1 shows the iSCSI targets for an Eight-RDX drive configuration.

CUSTOMIZING THE ISCSI DEVICES VIEW

You can sort the columns in ascending or descending order and select which columns are visible.

To customize the iSCSI Devices view:

Select the ▼ arrow, which appears when you move your mouse over the product or serial number column head.

| Physical Device | al Logical Device | | | |
|---|-------------------|-------------------------------------|---------------------|---------------------------|
| 🛃 Convert Device Type | | | 🔤 Move Med | ia 丨 💥 Advanced Options |
| 🗸 🌺 iSCSI Devices | iSCSI Devices | • | | |
| Disk Drive1 (Dock01) Disk Drive2 (Dock02) | Product Name | Serial Number - | iSCSI Name | |
| Disk Drive3 (Dock03) | RDX | Ag↓ Sort Ascending | iqn.2010-01.com.tar | ndbergdata:storage.rdx8-8 |
| Disk Drive4 (Dock04) | RDX | X ↓ Sort Descending | iqn.2010-01.com.tar | ndbergdata:storage.rdx3-3 |
| Disk Drive5 (Dock05) | RDX | | ign.2010-01.com.tar | ndbergdata:storage.rdx4-4 |
| Disk Drive7 (Dock07) | RDX | PD0016502602 | Product Name | xbergdata:storage.rdx6-6 |
| 📄 Disk Drive8 (Dock08) | RDX | PD0001752904 | Serial Number | bergdata:storage.rdx5-5 |
| | RDX | PD0000059008 | ISCSI Name | dbergdata:storage.rdx1-1 |
| | RDX | PD000000007 | iqn.2010-01.com.tar | ndbergdata:storage.rdx2-2 |
| | | | | |
| | | | | |
| | | | | |
| | 4 | | | N |
| | | | | |

Figure 5-2 Customizing the iSCSI Devices view

MANAGING DISK AND TAPE DRIVE CONFIGURATIONS

If you have your RDX QuikStation configured as either eight RDX drives or as eight virtual LTO-3 tape drives, you can do the following tasks from the Logical Device interface:

- View logical device properties (page 74)
- View active iSCSI connections (page 75)
- Set iSCSI security for each target (page 34)
- Convert the logical device type (page 48)
- **Note:** From the System Settings menu, you can set a removable disk option to allow unsafe eject of RDX cartridges. See page 56 for more information.

VIEWING LOGICAL PROPERTIES FOR RDX OR VIRTUAL TAPE DRIVES

The following properties are displayed for RDX and virtual LTO-3 tape drive targets.

| Property Name | Description |
|---|---|
| Vendor | The vendor name for the device. For RDX drives, the Vendor field always displays "TANDBERG." For virtual tape drives, the Vendor field displays "HP". |
| Product | The product name for the device. The Product field displays "RDX" for RDX drives and "Ultrium-3 SCSI" for tape drive configurations. |
| Revision | The firmware revision level for the logical device. |
| Serial number | The serial number for the logical device. |
| iSCSI name | The iSCSI name that identifies the RDX or tape drive target. The iSCSI name is used to connect the initiators (host computers) to the targets. |
| Target ID/LUN | The iSCSI target/logical unit number that identifies the RDX drive or tape drive. |
| Status | Displays the operating status for the device. The status displays either "good", "offline", or an error state. |
| Media label (visible if the dock has an RDX cartridge inserted) | The default media label is derived from the last eight digits of the RDX cartridge serial number, or the last six digits and "L3" for a tape. |
| Media state | The media state displays either "loaded" or "unloaded" depending on how the drive is set in your backup application. |
| Media usage | Note: Media usage for RDX drives is unavailable. This property is only reported for tape drive configurations. |
| | Displays the percentage of capacity that has been used for the RDX cartridge (or virtual tape cartridge). |

 Table 5-3
 Logical properties for RDX and LTO-3 tape drive targets

VIEWING ACTIVE ISCSI CONNECTIONS FOR RDX OR TAPE DRIVES

1. From the Logical Device interface, select an RDX or tape drive target from the tree view in the left pane.

Note:2.The Connection tab only displays when you have selected an iSCSI device target from the left pane.

| Physical Device | 😹 Logical Device | |
|---|---|--|
| 🛃 Convert Device Type | | 🔤 Move Media 🛛 💥 Advanced Options |
| Convert Device Type Disk Drive1 (Dock01) Disk Drive2 (Dock02) Disk Drive2 (Dock03) Disk Drive3 (Dock03) Disk Drive4 (Dock04) Disk Drive5 (Dock05) Disk Drive6 (Dock06) Disk Drive8 (Dock08) | Name Vendor Product Revision Serial Number iSCSI Name Target ID/LUN Status | Connection Value Value TANDBERG RDX 0063 PD0000059008 iqn.2010-01.com.tandbergdata:storage.rdx1-1.0025900(1/0 Good |
| | ٢ | |

3. Click the Connection tab.

If there is a host connected to the target, the host's initiator node name will be displayed in the Connection dialog window, as shown in the following example.

| 📱 Physical Device | 😹 Logical Device | | |
|---|-------------------------------------|------------|-----------------------------------|
| 🧟 Convert Device Type | | | 🛿 Move Media 丨 🎇 Advanced Options |
| iSCSI Devices Disk Drive1 (Dock01) Disk Drive2 (Dock02) Disk Drive3 (Dock03) Disk Drive4 (Dock04) Disk Drive5 (Dock05) Disk Drive6 (Dock06) Disk Drive7 (Dock07) Disk Drive8 (Dock08) | Connected Host IQN tandbergdata1 | Connection | Security |

Note: You can also set iSCSI security for each iSCSI target. See "Setting iSCSI Security in the Remote Management Console" on page 34 for more information.

MANAGING TAPE LIBRARY AND AUTOLOADER CONFIGURATIONS

If you have your RDX QuikStation configured as a virtual tape library or as a tape or disk autoloader, you can do the following tasks from the Logical Device interface:

- View logical properties (see page 76)
- View active iSCSI connections (page 77)
- Set iSCSI security for each target (page 34)
- Convert the logical device type (page 48)
- Logically move media to another slot (page 78)
- Set the Unique Inquiry String (page 80)
- Set the autoloader option to continuously rotate cartridges (only applicable to disk and tape autoloaders)
- **Note:** From the System Settings menu, you can set an option to allow auto-insert (import/export functionality) of media cartridges. See page 57 for more information.

VIEWING LOGICAL PROPERTIES FOR TAPE LIBRARIES/AUTOLOADERS

The following properties are displayed for virtual tape libraries and autoloaders.

| Property Name | Description |
|-------------------------------|--|
| For the library/autoloader ar | nd its tape drive(s) |
| Element address | The element address associated with the virtual location of the library/autoloader and drives. |
| Vendor | The vendor name for the virtual library/autoloader and drives. For example, the vendor name for a StorageLoader tape library is "Tandberg Data," and the vendor name for the tape drive is "HP." |
| Product | The product name for the virtual library/autoloader and drives. For example, the StorageLoader tape library product name is "StorageLoader". For the LTO-3 virtual tape drives, the product name is "Ultrium 3-SCSI." |
| Revision | The firmware revision level for the logical device. |
| Serial number | The serial number for the logical device. |
| iSCSI name | The iSCSI name that identifies the library/autoloader device and its drive targets. The iSCSI name is used to connect the initiators (host computers) to the targets. |

Table 5-4 Logical properties for virtual tape libraries and autoloaders

 Table 5-4
 Logical properties for virtual tape libraries and autoloaders

| Property Name | Description |
|---|--|
| Target ID/LUN | The iSCSI target/logical unit number that identifies the library/autoloader and drives. |
| Status | Displays the operating status for the device. The status displays either "good", "offline", or an error state. |
| For I/E port and tape slots | |
| Element Address | The element address associated with the virtual location of the tape library/autoloader slots. The element address correlates to the RDX dock number. |
| Media label (visible if the dock has an RDX cartridge inserted) | The default media label is derived from the last six digits of the RDX cartridge serial number and "L3", which refers to an LTO-3 tape cartridge. |
| Media state | The media state displays either "loaded" or "unloaded" depending on how the drive is set in your backup application. |
| Media usage (visible if the dock has an RDX cartridge inserted) | Note: Media usage for RDX drives is unavailable. This property is only reported for tape library, tape autoloader, and disk autoloader configurations. Displays the percentage of capacity that has been used for the RDX cartridge (or virtual tape cartridge). |

VIEWING ACTIVE ISCSI CONNECTIONS FOR TAPE LIBRARIES/AUTOLOADERS

- 1. From the Logical Device interface, select the library/autoloader or drive target from the tree view in the left pane.
 - **Note:** The Connection tab only displays when you have selected an iSCSI device target from the left pane.

| 📳 Physical Device 🛛 🔓 | Logical Device | | |
|--|---|---|-----------------------|
| 🛃 Convert Device Type | | 🔤 Move Media 💥 Advanced Optio | ns |
| Convert Device Type Convert Devices Convert Devices Convert Devices Convert Devices Convert Tape Drive1 (Dock01) Convert Tape Drive2 Convert Tape Drive2 Convert Tape Drive3 Convert Tape Drive4 Convert Tape Slot3 Convert Tape Slot4 Con | Properties Name Element Address Vendor Product Revision Serial Number iSCSI Name Target ID/LUN Status | Move Media X Advanced Optio Advanced Optio Security Value View currently connected hosts to the RE Only one active connection per logical de 97 TANDBERG StorageLoader 0418 90089E6AL110 iqn.2010-01.com.tandbergdata:storage.t40-1.00259008 1/0 Good | ns XX Qu Vice i |
| Slot4 (Dock04) Slot5 (Dock05) Slot6 (Dock06) Slot7 (Dock07) Slot8 (Dock08) | < | | 9 |

2. Click the Connection tab.

If there is a host connected to the target, the host's initiator node name will be displayed in the Connection dialog window, as shown in the following example.

| 📱 Physical Device 🛛 📕 | Logical Device | | | |
|-----------------------|--------------------|------------|--------------|--------------------|
| 🛃 Convert Device Type | | | 🔤 Move Media | 💥 Advanced Options |
| Storagel ibrary T40 | Properties | Sonnection | 🤯 Security | |
| Tape Drive1 (Dock01) | Connected Host IQN | | | |
| Tape Drive2 | tandbergdata1 🦳 | | | |
| Tape Drive3 | | | | |
| Tape Drive4 | | | | |
| IE Port | | | | |
| 4 🚰 Tape Slots | | | | |
| 🔤 Slot1 | | | | |
| 🔤 Slot2 (Dock02) | | | | |
| 🔤 Slot3 (Dock03) | | | | |
| 🔤 Slot4 (Dock04) | | | | |
| 🔤 Slot5 (Dock05) | | | | |
| 🔤 Slot6 (Dock06) | | | | |
| 🔤 Slot7 (Dock07) | | | | |
| 🔤 Slot8 (Dock08) | | | | |
| | | | | |
| | | | | |
| | L | | | |

Note: > You can also set iSCSI security from the iSCSI Management dialog window. See "Setting iSCSI Security in the Remote Management Console" on page 34 for more information.

MOVING A MEDIA CARTRIDGE

With RDX QuikStation virtual library and tape autoloader configurations, you can logically move media to an open slot, to the IE port, or to the tape drives.

Note: If you move media to the IE port, the cartridge will be ejected from the dock, and the media will be logically removed from the library.

In most cases, you should use your backup software application to logically move media within the virtual library or autoloader. This functionality is built into the RDX QuikStation mainly for testing and diagnostic purposes.

Important Make sure that you do not have any backup jobs in progress before you logically move a cartridge.

To logically move media:

- 1. From the Logical Device interface, select the tape slot, tape drive, or IE Port from the tree view in the left pane.
- 2. Select Move Media.

| 📳 Physical Device 🛛 🛃 | Logical Device | |
|--|-----------------|--------------|
| 🛃 Convert Device Type | | Move Media |
| 4 🍓 iSCSI Devices | Properties | 40 |
| StorageLibrary T40 Tape Drive1 | Name | Value |
| Tape Drive2 | Element Address | 1 |
| Tape Drive3 | Media Label | 7A007AL3 |
| III Tape Drive4 III Port ▲ 🍓 Tape Slots | Media Usage | 0/750GB (0%) |
| Slot1 (Dock01) Slot2 (Dock02) Slot3 (Dock03) | | |
| Slots (Dock07) Slots (Dock05) Slots (Dock06) Slot7 (Dock07) | | |
| SIGG (DOCK00) | | |

The Move Media dialog window opens.

| Move Media | | × |
|--|---|---|
| This function ma connected to th the current back encountered, pe backup application | y conflict with any backup application e logical library and could result in a failure of up or restore operation. If a failure is rform a re-inventory operation from your on and try the job again. | |
| Source: | Slot1 | |
| Destination: | ~ | |
| | Move | |

- 3. Select the Source slot and the Destination slot from the pull-down menus. The Source pull-down menu only displays the slots that contain an inserted cartridge, and the Destination pull-down menu only displays empty slots.
- 4. Click Move.

SETTING THE UNIQUE INQUIRY OPTION

The Unique Inquiry option is available from the Advanced Options dialog.

You can set the RDX QuikStation to report the tape library/autoloader's unique inquiry string, or an RDX QuikStation inquiry string

 Table 5-5
 Inquiry String settings

| Inquiry String setting | Behavior |
|---|--|
| Original device inquiry string (Default) | The virtual tape library reports the device's original inquiry string. This is the inquiry string that a physical version of the tape library reports. For example, the device inquiry string for a Tandberg Data StorageLibrary T24 is <i>Magnum 224</i> (derived from the library's former name). |
| RDX QuikStation inquiry string | The virtual tape library reports "RDX QuikStation" as the device inquiry string. This option is mainly used for diagnostic purposes. |

SETTING THE AUTOLOADER OPTION

Note: The autoloader option does not apply to tape library emulation modes. It is only available for disk autoloader or tape autoloader configurations.

The following table describes RDX QuikStation behavior when this option is turned off and when it's turned on.

In cases where you do not want the cartridge to physically eject from the dock, you can set the option to have cartridges continuously rotate through the slots. This option eliminates the need to re-insert the cartridges into the RDX docks.

| Continuously rotate cartridges in the slots | Behavior |
|---|---|
| Off (Default) | Physically ejects the cartridge from the RDX dock when the virtual drive completes a backup job. Once the cartridge is ejected, the virtual autoloader loads media from the next subsequent slot into the drive. It will load cartridges from the slots sequentially and will stop when it reaches the last slot, even if new cartridges have been inserted. |
| On (checkbox is selected) | Does not physically eject the cartridge from the RDX dock when the virtual drive completes a backup job. Instead, the cartridges continuously rotate sequentially through the slots. |

 Table 5-6
 Autoloader option settings

NOTES

6

PHYSICAL DEVICE MANAGEMENT

This chapter describes the Remote Management Console's Physical Device interface.

The physical device view is useful for determining the physical location of your RDX cartridges, since logical tape library views can differ. You can also use the physical device view to force-eject RDX cartridges, erase cartridges, and clone data from one RDX cartridge to another.

The chapter includes the following sections:

| | Viewing physical device properties pag | ;e 84 |
|----|---|-------|
| | Ejecting RDX cartridges pag | ;e 86 |
| | Erasing RDX cartridges pag | ;e 86 |
| | Cloning data | ;e 87 |
| No | e: For information on performing a dock diagnostic test, see Chapter page 95. | 7 on |

VIEWING PHYSICAL DEVICE PROPERTIES

| Physical Device | 4 | Logical Device | | |
|---|------------|-------------------|---|----|
| | | | 📤 Eject 🧷 Erase / Clone 🕮 Tr | BS |
| ⊿ 🥝 QS9890 | | US9890 Properties | | |
| Dock01 (7A007AL3) Dock02 (127BR9L3) Dock03 (G0N0KAL3) Dock04 (7A00EAL3) Dock05 (B25CSWL3) | | Name | Value | |
| | | System Status | Good | |
| | | System Temp(C) | 32degree | |
| | | Host Name | QS9890 | |
| Dock06 (601391L3) Dock07 (Empty) Dock08 (58818013) | 1L3) | Domain | td.tandbergdata.com | |
| |) 01.2) | IP Address | 10.1.50.211 (Static) | |
| DOCK08 (28818913) | | Link Speed | 1Gbps | |
| | | Serial Number | 98900000 | |
| | | Version | 02.010.00 | |
| | | DOM Version | 1.6 | |
| | | ADM Version | 1.2 | |
| | | | | |

The following physical device properties are displayed when you select the RDX QuikStation's host name in the left pane of the Physical Device interface.

| Property Name | Description |
|-----------------------|---|
| System status | The operating system status for the RDX QuikStation. The status can either display "good", "offline", or an error state. |
| System temp (Celsius) | The system temperature for the RDX QuikStation. For information on the RDX QuikStation's operating temperature range, see page 106. |
| Host name | The name that identifies the RDX QuikStation. The default host name is "QS" plus the last four digits of the RDX QuikStation serial number. You can change the host name from the Network Configuration dialog window. See page 50 for more information. |
| Domain | The search domain name for the network where the RDX QuikStation resides. This property field is populated from the Network Configuration dialog window. If a search domain is not configured, this field will be blank. See page 50 for more information. |
| Link speed | The network connection link speed. |
| Serial number | The RDX QuikStation product serial number. |
| Version | The firmware version that is currently running on the RDX QuikStation. |

 Table 6-1
 RDX QuikStation physical device properties

| Property Name | Description |
|---------------|---|
| DOM version | The Disk On Module (DOM) version that is currently in use. |
| ADM version | The Admiral board (ADM) firmware version that is currently in use. The Admiral board enables communication to the center-front LED panel, dock LCD panels, fans, and system sensors. |

 Table 6-1
 RDX QuikStation physical device properties (continued)

The following RDX dock properties are displayed when you select an RDX dock in the left pane of the Physical Device interface.

Table 6-2RDX dock properties

| Property Name | Description | |
|------------------------------|---|--|
| Dock status | Displays the operating status for the RDX dock. The status displays either "good", "offline", or an error state. For information on running a dock diagnostic test, see page 95. | |
| Dock LCD text | The cartridge label text that is displayed on the LCD dock panel. This field is populated only if an RDX cartridge is inserted in the dock. | |
| Dock FW rev | The RDX dock firmware version. | |
| With a cartridge inserted in | the dock: | |
| Cartridge status | Displays the operating status for the RDX cartridge. The status displays either "good" or an error state. | |
| Cartridge serial number | The product serial number for the RDX cartridge. | |
| Cartridge label | By default, the text is derived from the RDX cartridge serial number. | |
| Cartridge capacity | The RDX cartridge's total capacity (used and unused). | |
| Cartridge temp (Celsius) | The current temperature of the inserted RDX cartridge. | |
| Write-protected | Displays information on whether the RDX cartridge is write-protected or not. | |

EJECTING RDX CARTRIDGES

You can eject an RDX cartridge from the Remote Management Console. However, with removable disk configurations, you cannot eject a cartridge that is connected to a host computer unless you first enable the Unsafe Eject option (see page 76).

Important Make sure that you do not have any backup jobs in progress before you eject an RDX cartridge from the Remote Management Console.

To eject an RDX cartridge:

1. From the Physical Device interface, select the dock where the cartridge is located.

| QS9890 | Cock 01 Properties | V |
|----------------------------|-------------------------|---------------|
| Dock01 (7A007AL3) | Name | Value |
| Dock02 (1278K9L3) | Dock Status | Good |
| Dock04 (7A00EAL3) | Dock LCD Text | 7A007AL3 |
| EDock05 (B25CSWL3) | Dock FW Rev | 0063 |
| M Dock06 (601391L3) | Cartridge Status | Good |
| Dock07 (Empty) | Cartridge Serial Number | 575466F0 |
| - DOCKO8 (28818913) | Cartridge Label | 7A007AL3 |
| | Cartridge Capacity | 750GB |
| | Cartridge Temp(C) | 23C degrees |
| | Write Protected | Not protected |

The RDX QuikStation ejects the cartridge. The dock displays "Empty" in the tree view.

ERASING RDX CARTRIDGES

You can erase data and reformat an RDX cartridge as a single NTFS partition from the Remote Management Console.

If you configure your RDX QuikStation as a virtual tape library, the RDX cartridges are automatically formatted to use with the library. This format partitions 8 MB as an NTFS file system. The rest of the cartridge's capacity is formatted to emulate a tape cartridge.

If you configure your RDX QuikStation as eight removable disks, the RDX cartridges should be formatted as a single NTFS file partition. You can reformat a cartridge from the Remote Management Console. If the disk is connected to a host computer, you can also reformat the cartridge from the host.

To erase an RDX cartridge:

1. From the Physical Device interface, select the dock where the cartridge is located.

| | | Eiert Zerase Toole Cone |
|--|-------------------------|-------------------------|
| 🔮 QS9890 | Cock 01 Properties | Û |
| Dock01 (7A007AL3) Dock02 (127BR9L3) Dock03 (G0N0KAL3) Dock04 (7A00EAL3) Dock05 (B25CSWL3) Dock06 (601391L3) | Name | Value |
| | Dock Status | Good |
| | Dock LCD Text | 7A007AL3 |
| | Dock FW Rev | 0063 |
| | Cartridge Status | Good |
| Dock07 (Empty) | Cartridge Serial Number | 575466F0 |
| Dock08 (588189L3) | Cartridge Label | 7A007AL3 |
| | Cartridge Capacity | 750GB |
| | Cartridge Temp(C) | 23C degrees |
| | Write Protected | Not protected |

- 2. Click Erase.
- 3. Click Yes to confirm.

CLONING DATA

You can copy data from one RDX cartridge to another by using the cloning feature in the Remote Management Console.

Note: The cloning function copies the source cartridge to a destination cartridge, sector by sector, so that the destination cartridge will be an exact image of the source cartridge. Therefore, cloning will be rejected if the destination cartridge is smaller than the source cartridge.

To clone data:

- 1. Insert both the source RDX cartridge and the destination RDX cartridge into an available dock on the RDX QuikStation.
- 2. From the Physical Device interface, click Clone.

| Dock01 (7A007AL3) | Name | Value |
|--------------------------|----------------|----------------------|
| Dock03 (G0N0KAL3) | System Status | Good |
| Dock04 (7A00EAL3) | System Temp(C) | 32degree |
| Dock05 (B25CSWL3) | Host Name | QS9890 |
| Dock06 (601391L3) | Domain | td.tandbergdata.com |
| Dock07 (Empty) | IP Address | 10.1.50.211 (Static) |
| Dock08 (588189L3) | Link Speed | 1Gbps |
| | Serial Number | 989000000 |
| | Version | 02.010.00 |
| | DOM Version | 1.6 |
| | | 4.0 |

The Clone dialog window opens.

| Clone | | × |
|---|--|---|
| IIIIATTENTIONIIII during the cloning the source cartric a long time to co Also be aware th destination cartrid | destination RDX cartridges are offline g process. Depending on the size of Ige, the cloning process could take mplete. at the current data on the Ige will be lost once cloning starts. | |
| Source: | ~ | |
| Destination: | ¥ | |
| | Start Cancel | |

- 3. Select the Source RDX cartridge from the pulldown menu.
- 4. Select the Destination RDX cartridge from the pulldown menu.
- **Note:** The source and destination cartridges will be offline during the cloning process.
- 5. Click Start.

The RDX QuikStation begins the cloning process. You can monitor the progress of the cloning job in the bottom-right corner of the Physical Device interface.

| rck QUIKSTAT | | | TANDBERG DATA |
|---|---|---------------------------------------|--|
| Welcome, ADMIN | | 🍪 System Settings 🔹 🎤 System Maint | enance 🔹 🌏 Refresh 🛛 📕 Sign Out 🛛 🔞 Help |
| UTADO7A | T7127BR9 | CLGONOKA 907A00EA | |
| Physical Device | Jugical Device | 📥 Eject 🧷 E | rase 🚽 Clone 🇀 Test |
| ✓ Q Q S9890 ■ Dock01 (T5B25CSW) ■ Dock02 (T7127BR9) ■ Dock03 (QLG0N0KA) ■ Dock04 (907A00EA) ■ Dock05 (0L7A007A) ✓ Dock05 (0L7A007A) ✓ Dock05 (0CF6906) ✓ Dock08 (Offline) | Mame Dock 06 Properties Dock Status Dock LCD Text Dock FW Rev | Value Offline -OFFLINE- 0063 | |
| Cloning Dock06 to Dock08 | | 6% c | ompleted Cancel |

[∖]cloning job status bar

NOTES

7

System Maintenance

This chapter describes the RDX QuikStation system maintenance tasks that a user with administrator privileges can perform.

The chapter includes the following sections:

| System maintenance overview page 92 |
|--|
| Performing an LCD diagnostic test page 93 |
| Performing a dock diagnostic test page 95 |
| Generating system logs page 96 |
| • Restarting and shutting down the RDX QuikStation page 97 |
| • Exporting system configuration files page 98 |
| Importing system configuration files page 98 |
| Manually updating firmware page 99 |
| Using a USB flash drive for system tasks page 101 |

SYSTEM MAINTENANCE OVERVIEW

As a user with administrator permissions, the following functions are available for you to perform from the Management Console's System Maintenance menu.

| System Maintenance Menu Name | Description | |
|---------------------------------|--|--|
| LCD Diagnostic Test | Allows you to verify that all eight LCD docks and the LED center panel are working properly. You must be physically in front of the RDX QuikStation to monitor the test sequence. See page 93 for more information. | |
| System Log | Allows you to download system log files to a host computer. These files are encrypted and are meant for Tandberg Data technical support only. | |
| System Restart | Allows you to restart the RDX QuikStation. | |
| | Important: Before you restart the system for any reason, make sure there are no active host connections to the iSCSI targets. | |
| System Shutdown | Allows you to perform a soft shutdown of the RDX QuikStation. A soft shutdown is the preferred way of turning off the power to the unit. | |
| | If you need to do a hard shutdown, press and hold the power button on the back of the RDX QuikStation for 10 seconds. | |
| | Important: Before you shutdown the system for any reason, make sure there are no active host connections to the iSCSI targets. | |
| Import Configuration | Allows you to import RDX QuikStation system configuration files, such as network information, email notification setting, time zone, user names and password logical device type settings, and iSCSI security settings to another RDX QuikStation. | |
| Export Configuration | Allows you to export RDX QuikStation system configuration settings, such as network information, email notification, time zone, user names and passwords, logical device type settings, and iSCSI security settings. | |
| | Tandberg Data recommends that you export your system configuration files as a backup in case you ever need to replace your current unit. | |
| Update Firmware | Allows you to automatically check for updates and manually upload a firmware file. | |
| | Important: Before you update firmware, make sure there are no active host connections to the iSCSI targets. | |

 Table 7-1
 RDX QuikStation System Maintenance Menu

PERFORMING AN LCD DIAGNOSTIC TEST

An LCD diagnostic test verifies that the LCD/LED pixels and backlighting are working correctly. You must be standing in front of the RDX QuikStation to verify the test sequence.

To run an LCD diagnostic test:

1. From the Remote Management Console main menu, select System Maintenance > LCD Diagnostic Test.



2. Click Yes.

The test will begin in five seconds.

3. Stand in front of the RDX QuikStation to verify that the following test sequence (see page 94) occurs.

LCD DIAGNOSTIC TEST SEQUENCE

The test begins with dock 1 then moves sequentially through each dock, ending with dock 8.



The LCD diagnostic test sequence is as follows:

- 1. At dock 1, the LCD panel pixels all turn off, then all turn on.
- 2. The dock's power LED/eject button alternately flashes amber and green. If a cartridge is loaded in the dock, the cartridge's activity LED also alternately flashes amber and green.
- 3. The dock's LCD panel backlight flashes, then turns off.
- 4. The test moves to the next dock, completing the same sequence described in steps 1 through 3 for each subsequent dock.
- 5. When the LCD test for each dock has completed, the error indicator and the USB indicator flash on the LED center panel.
- 6. All eight dock LCD panels flash "DONE" when the test is complete.

PERFORMING A DOCK DIAGNOSTIC TEST

You do not have to be physically in front of the RDX QuikStation to conduct a dock diagnostic test. However, you do need an RDX cartridge in the dock to verify that the dock is working properly.

To perform a dock diagnostic test:

- 1. Select the Physical Device tab in the Remote Management Console.
- 2. Select the dock that you want to test, then click <u>Dock Diagnostic Test</u> in the upper-right corner of the Physical Device interface.

| | | Eiect Perase Dock Diagnos |
|---|-----------------------|---------------------------|
| © QS9890 Dock01 (601391L3) Dock02 (505957L3) Dock03 (283628L3) Dock04 (Empty) Dock05 (Empty) Dock05 (Empty) | Dock 01 Properties | |
| | Name | Value |
| | Dock Status | Good |
| | Dock LCD Text | 601391L3 |
| | Dock FW Rev | 0063 |
| Dock07 (505996L3) | Cartridge Status | Good |
| Dock08 (Empty) | Cartridge Serial Numl | 859900601391 |
| | Cartridge Label | 601391L3 |
| | Cartridge Capacity | 160GB |
| | Cartridge Temp(C) | 24C degrees |
| | Write Protected | Not protected |

The test begins. If the dock is working properly, the following dialog window is displayed.

| Fest Passed X | |
|--|---|
| The dock and the cartridge are working properly. | |
| OK | l |
| | |

- 3. Click OK.
- **Note:** If the test encounters a problem, the system displays an error message. For more information, see Appendix A for a list of error codes, or contact Tandberg Data technical support.

GENERATING SYSTEM LOGS

There are three levels of system logs that you can generate from the RDX QuikStation: Normal, Detail, and Extend (see page 54).

System logs are encrypted and only intended for use by Tandberg Data technical support. You do not need to generate a system log unless you are instructed to do so by Tandberg Data technical support.

To generate a system log:

 From the Remote Management Console main menu, select System Maintenance > System log.



- 2. Save the file to your local host.
- **Note:** You can also generate a log dump by inserting a USB flash drive (fomatted with a VFAT file system) into the RDX QuikStation. The RDX QuikStation automatically generates a log dump when a USB flash drive is detected. Make sure that the flash drive does not include any RDX QuikStation system configuration files (see page 101). Otherwise, the RDX QuikStation will read those files instead of generating a system log.

RESTARTING AND SHUTTING DOWN THE RDX QUIKSTATION

Important Before you restart or shutdown the system for any reason, make sure there are no active hosts that are currently communicating with the iSCSI targets.

To restart the RDX QuikStation:

 From the Remote Management Console main menu, select System Maintenance > System Restart.



2. Click Yes to confirm.

The system reboots. This process takes three minutes. When the reboot is complete, the Remote Management Console interface is active again. You do not have to sign in again to access the Console.

To shutdown the RDX QuikStation:

 From the Remote Management Console main menu, select System Maintenance > System Shutdown.



- 2. Click Yes to confirm.
- **Note:** If you cannot access the Remote Management Console and need to shutdown, press and hold the power button on the back of the RDX QuikStation for 10 seconds.

EXPORTING SYSTEM CONFIGURATION FILES

As a best practice, Tandberg Data recommends that you export your system configuration files as a backup in case you ever need to replace your current unit. The system exports all configuration settings, including network information, email notification settings, time zone setting, user names and passwords, logical device type settings, and iSCSI security settings.

To export RDX QuikStation system configuration files:

1. From the Remote Management Console main menu, select System Maintenance > Export Configuration.



The Export Configuration dialog window opens.



- 2. Click Export to generate the configuration file.
- 3. Click Ok to download the file to your local host.

IMPORTING RDX QUIKSTATION SYSTEM CONFIGURATION FILES

To import system configuration files:

1. From the Remote Management Console main menu, select System Maintenance > Import Configuration.


The Import Configuration dialog window opens.

| Import Configuration | × | |
|--|--------|--|
| Select the configuration settings that you want to import to the RDX QuikStation. In some cases, the system restarts after the import completes. | | |
| Configuration to import Network Settings Include identification information (host name, static IP address) | | |
| Time Zone/Time Server Vser Account Settings LCD Language Notification Settings | | |
| Configuration File: Select a file. | Cancel | |

- 2. Select the network settings you want to import. By default, all configuration settings are selected.
- 3. Click the + icon to select the RDX QuikStation configuration file.
- 4. Click Import.

The system reboots. This process takes three minutes. When the reboot is complete, the Remote Management Console interface is active again. You do not have to sign in again to access the Console.

MANUALLY UPDATING FIRMWARE

When you sign in as an administrator user, the RDX QuikStation detects the firmware version (see page 54). If your system is not running the most current version, the RDX QuikStation prompts you to update the firmware. If you choose to update, the system automatically installs to the new firmware level.

Follow these instructions if you need to manually update the firmware file to your system.

Important Before you update firmware, make sure there are no active hosts that are currently communicating with the iSCSI targets.

To manually update the RDX QuikStation firmware:

 From the Remote Management Console main menu, select System Maintenance > Update Firmware.



The Update Firmware dialog window opens.

| Up | date Firmware X | Ì |
|----|---|---|
| | Current Firmware Level: 01.014 Select an option from below: O Check for updates automatically Update QuikStation firmware manually | |
| | Continue Cancel | |

2. Select the "Update QuikStation firmware manually" radio button.

| Update Firmware | | × |
|---|---|---|
| If you do not have <u>web site</u> . | a firmware file, you may download one from the <u>Tandberg Data Support</u> | |
| Firmware File: | vtx-01.015.fw | |
| | Update | |

3. Click the "Tandberg Data Support Web Site" link to access the most current firmware file.

After you download the file from the web site, make sure you extract the zipped file before uploading it to the RDX QuikStation. The firmware file uses the following format: vtx-<version>.fw.

- 4. Click the 🕂 icon to select the RDX QuikStation configuration file.
- 5. Click Update].
- 6. Click Yes to proceed.

The system reboots. This process takes three minutes. When the reboot is complete, the Remote Management Console interface is active again. You do not have to sign in again to access the Console.

Note: You can also update firmware by saving the firmware file to a USB flash drive (fomatted with a VFAT file system). Rename the firmware file "**vtx.fw**" and insert the flash drive into the RDX QuikStation. The RDX QuikStation automatically detects the firmware and updates the system.

USING A USB FLASH DRIVE FOR SYSTEM TASKS

The RDX QuikStation ships with a USB flash drive that you can use to perform many system tasks. If the RDX QuikStation flash drive is unavailable, you can use any USB flash drive that is formatted with a VFAT file system.

Important The Tandberg Data-supplied USB flash drive includes the file **network.conf**. If you are using the flash drive for a system task other than network configuration, make sure that this file name is changed to another name, for example **network.conf.used**, before you insert the flash drive into the RDX QuikStation.

Renaming the file prevents any inadvertent network configuration changes to your RDX QuikStation.

The following table describes the system tasks you can complete with a USB flash drive.

| System Task | Description |
|--------------------------|---|
| Network configuration | You can configure network settings for the RDX QuikStation via the network.conf file, which is included on your RDX QuikStation USB flash drive. When the flash drive is inserted into the RDX QuikStation, the system automatically detects and imports the network settings from the file. For more information, see "Configuring Network Information" on page 18. |
| | Important: The file must be named network.conf in order for the RDX QuikStation to import network settings. |
| Firmware updates | You can update firmware by saving the firmware file to a USB flash drive and inserting the flash drive into the RDX QuikStation. The RDX QuikStation automatically detects the firmware and updates the system. For more information, see "Manually Updating Firmware" on page 99. |
| | Important: Rename the firmware file " vtx.fw " so that the RDX QuikStation can detect and load the file. |
| Password recovery | You can reset the built-in administrator password via USB by creating a password.reset file. For more information, see "Recovering the Built-In Administrator Password" on page 64. |
| System log dump | You can generate a log dump by inserting a USB flash drive into the RDX QuikStation. The RDX QuikStation automatically generates a system log when a USB flash drive is detected. For more information, see "Generating System Logs" on page 96. |
| | Important: If the following files – network.conf, password.reset, or vtx.fw – are loaded onto the flash drive, the RDX QuikStation will not generate a log dump, but will instead read and load the file. |

 Table 7-2
 Uses for RDX QuikStation USB Flash Drive

NOTES

A

TROUBLESHOOTING

This appendix provides information on error codes and some basic troubleshooting questions and solutions.

For more detailed troubleshooting help, visit the Support section of the Tandberg Data web site (www.tandbergdata.com), or contact Tandberg technical support.

RDX QUIKSTATION DOCK ERROR CODES

The following error codes will display if there is a fault condition associated with the RDX dock.

| Error code | Meaning | Potential Solution |
|------------|---|---|
| 01 | POST failed. This is an internal system error associated with the RDX dock. | Reboot the RDX QuikStation (see page 92). |
| 02 | Link down. A communication link error between the RDX cartridge and RDX dock. | Eject and re-insert the RDX cartridge. If the error persists, reboot the system (see page 92) |
| 03 | Internal system error associated with the RDX dock. | Reboot the RDX QuikStation (see page 92). |
| 04 | Eject failed. The RDX cartridge failed to eject from the dock. | Retry ejecting the cartridge. If the error persists, reboot the system (see page 92). If the error still persists, eject the cartridge via the emergency-eject keyhole (see page 4). |
| 05 | Firmware image bad. The RDX dock firmware should be reinstalled. | Contact Tandberg Data technical support. |
| 06 | Communication error between the RDX QuikStation and the RDX dock. | Reboot the RDX QuikStation (see page 92). |

Table A-1Error codes

| Table A-1 | Error codes | (continued) |
|-----------|-------------|---------------------------------------|
| | | · · · · · · · · · · · · · · · · · · · |

| Error code | Meaning | Potential Solution |
|------------|---|---|
| 07 | Offline. The RDX dock is offline and unavailable for use. | Reboot the system (see page 92). Important: The RDX QuikStation docks go offline when performing cloning or diagnostic operations. Make sure your system is not performing diagnostics before you reboot the RDX QuikStation. |
| 08 | Unknown error. The RDX dock has an unknown internal system error. | Reboot the system (see page 92). If the error persists, contact Tandberg Data technical support. |

BASIC TROUBLESHOOTING INFORMATION

The following table describes some basic troubleshooting information. If you have a problem that is not addressed in the table below, contact Tandberg Data technical support.

 Table A-2
 Basic troubleshooting information

| Problem | Potential Solution(s) |
|--|--|
| Can't connect to an iSCSI target. | Check to see if another host is already connected to the target (see page 75 and page 77). The RDX QuikStation allows only one active host connection per iSCSI target (see page 24). |
| | Check if any iSCSI security is enabled for the target (see page 34). |
| | If the target uses CHAP authentication, make sure the CHAP secret specified in the Remote Management Console (see page 35) matches the CHAP secret specified in your iSCSI initiator software. |
| | • Refresh the discovered targets in your iSCSI initiator software. |
| The Remote Management Console has a "session timeout" error | The Remote Management Console times out after 30 minutes of inactivity. |
| message. | Click OK to return to the Console's sign-in page. Type your user name and password to re-connect to the Remote Management Console (see page 20). |
| RDX dock reports as "empty" in the Remote Management Console when an RDX cartridge is loaded in the dock. | The RDX cartridge may not be seated properly in the dock. Push the cartridge into the dock until you hear it click in place. |

B

This appendix provides the following information about the RDX QuikStation:

| Physical specificationspage 10 | 06 |
|--|----|
| Power specificationspage 10 | 06 |
| Environmental specificationspage 10 | 06 |
| Safety and regulatory agency compliancepage 10 | 07 |

PHYSICAL SPECIFICATIONS

| Part Number | 8900-RDX | |
|---------------------------------------|--|--|
| Dimensions (D x W x H) | 508mm x 438mmx 89mm/ 20" x 17.25" x 3.5" | |
| Weight | 10.16 kilograms/22.4 pounds | |
| Form Factor | 2U Rackmount | |
| Capacity | Cartridge dependent (online): 8 x 1TB RDX cartridges | |
| | Unlimited offline capacity | |
| Data Transfer Rate | Up to 200 MB ^a /second across multiple docks (uncompressed) | |
| Network Connectivity | 2 x 1 Gigabit Ethernet, bonded for port failover | |
| USB 2.0 Ports | 1 on front panel, 2 on rear panel | |
| Removable Media Drive | 8 x RDX bays with individual LCD display | |
| RDX Media Compatibility (Capacity) | Backward and forward: 160GB, 320GB, 500GB, 640GB, 750GB, and 1TB | |

^a The data transfer rate is up to 200 MB/second when port configuration is configured for adaptive load balancing (see page 51).

POWER SPECIFICATIONS

| AC Input Voltage | 100-240 VAC |
|-----------------------|--------------------|
| Rated Input Current | 2,0 -1,0 A |
| Rated Input Frequency | 50/60 Hz |
| Power Consumption | <90 Watts, typical |

ENVIRONMENTAL SPECIFICATIONS

| Operating Temperature | 10° to 40° C (50° to 104° F) |
|---------------------------|-------------------------------|
| Non-operating Temperature | -20° to 60° C (-4° to 140° F) |
| Operating Humidity | 20% to 80% |

SAFETY AND REGULATORY AGENCY COMPLIANCE

Tandberg Data products are designed, tested, and classified for their intended electromagnetic environment. The RDX QuikStation is a Class A device and complies with the safety and regulatory agency standards listed below when installed in accordance with this manual.

Note: For regulatory purposes, the RDX QuikStation is filed as regulatory model number RMN-A-05-10.

UNDERWRITERS LABORATORY



The RDX QuikStation is listed by Underwriters Laboratories, Inc. Representative samples of this product have been evaluated by UL and meet the applicable U.S. and Canadian safety standards. UL60950-1:2007, Second Edition

CSA C22.2 No. 60950-1 Second Edition, 2007

GS



The GS Mark for the RDX QuikStation demonstrates that it has met the requirements of EN60950-1/A1:2010.

EUROPEAN COMMUNITY



This Information Technology Equipment has been tested and found to comply with the following European directives:
(1) EMC Directive 2004/108/EEC, according to:
EN55022: 2006 +A1:2007.CSPR22:2008
Warning: This is a class A product. In a domestic environment, this product may cause radio interference in which case the user may be required to take adequate measures.
EN61000-3-2:2006
EN61000-3-2:2006
EN61000-3-3:1995 +A1:2001 +A1:2005
EN55024:1998 +A1: 2001 +A2:2003
EN61000-6-2:2005
(2) Low Voltage Directive 2006/95/EEC

UNITED STATES: FCC DECLARATION OF CONFORMITY



We declare under our sole responsibility that: Product Name: RDX QuikStation, regulatory model RMN-A-05-10 To which this declaration relates, is in conformity with the

following standard(s) or other normative documents:

ANSI C63.4-1992 Methods of Measurement

Federal Communications Commission 47 CFR Part 15, Subpart B, Class AB:2008

This equipment has been tested and found to comply with the limits for a Class AB digital device, pursuant to Part 15 of the FCC Rules. These limits are designed to provide reasonable protection against harmful interference when the equipment is operated in a commercial environment. This equipment generates, uses, and can radiate radio frequency energy and, if not installed and used in accordance with this manual, may cause harmful interference to radio communications.

Operation of this equipment in a residential area is likely to cause harmful interference in which case the user will be required to correct the interference at his own expense.

CANADIAN VERIFICATION



This Class AB digital apparatus complies with ICES-003:2004.

Cet appareil numérique de la classe AB est conforme à la norme NMB-003 du Canada.

JAPAN



この装置は、クラスA情報技術装置です。この装置を家庭環境で使用する と電波妨害を引き起こすことがあります。この場合には使用者が適切な対策 を講ずるよう要求されることがあります。 VCCI-A

Translation: This is a Class A product based on the standard of the Voluntary Control Council for Interference by Information Technology Equipment (VCCI). If this equipment is used in a domestic environment, radio disturbance may arise. When such trouble occurs, the user may be required to take corrective actions.

AUSTRALIA AND NEW ZEALAND



This device has been tested and found to comply with the limits for a Class A digital device, pursuant to the Australian/New Zealand standard AS/NZS 3548:1995: "Information Technological Equipment" set out by the Australian Communications Agency.

WASTE OF ELECTRONIC AND ELECTRICAL EQUIPMENT (WEEE)



This device is in compliance with 2002\96\EC Waste of Electronic and Electrical Equipment (WEEE).

CHINA ROHS



This product is marked in accordance with China's Ministry of Information Industry (MII) "Management Methods for Controlling Pollution Cause by Electronic Information Products" Order #39. The Environmental Protection Use Period (EPUP) is determined to be 10 years. Packaging materials are also marked according to Chinese national standard GB 18455 - 2001 "Packaging and Recycling Symbols".

NOTES

C SAFETY GUIDELINES

This appendix provides safety guidelines for the RDX QuikStation:

| General safety precautions | . page 112 |
|-------------------------------|------------|
| Electrical safety precautions | page 113 |
| • ESD precautions | page 113 |
| • Operating precautions | . page 114 |

GENERAL SAFETY PRECAUTIONS

- Keep the area around the RDX QuikStation clean and free of clutter.
- Remove any jewelry or metal objects from your body, which are excellent metal conductors that can create short circuits and harm you if they come in contact with printed circuit boards or areas where power is present.
- Elevated Operating Ambient If installed in a closed or multi-unit rack assembly, the operating ambient temperature of the rack environment may be greater than room ambient. Therefore, consideration should be given to installing the equipment in an environment compatible with the maximum ambient temperature (Tma) specified by the manufacturer (35° C or 95° F).
- Reduced Air Flow Installation of the equipment in a rack should be such that the amount of air flow required for safe operation of the equipment is not compromised.
- Mechanical Loading Mounting of the equipment in the rack should be such that a hazardous condition is not achieved due to uneven mechanical loading.
- Circuit Overloading Consideration should be given to the connection of the equipment to the supply circuit and the effect that overloading of the circuits might have on overcurrent protection and supply wiring.
 Appropriate consideration of equipment nameplate ratings should be used when addressing this concern.
- Reliable Earthing Reliable earthing of rackmounted equipment should be maintained. Particular attention should be given to supply connections other than direct connections to the branch circuit (e.g. use of power strips).
- Never place objects on top of rackmounted equipment.

ELECTRICAL SAFETY PRECAUTIONS

- Be aware of the locations of the on/off switch on the chassis as well as the room's emergency power-off switch, disconnection switch or electrical outlet(s). The on/off switch does not disconnect power to the chassis. If an electrical accident occurs, quickly remove power to the system by removing the plug(s) from the outlet(s). Some models may have multiple power cords which connect to more than one outlet.
- The power cord must include a grounded plug and must be plugged into a grounded electrical outlet.
- Use only one hand when working with powered-on electrical equipment. This is to avoid making a complete circuit, which will cause electrical shock. Use extreme caution when using metal tools, which can easily damage electrical components or circuit boards they come into contact with.

ESD PRECAUTIONS

Electrostatic discharge (ESD) is generated by two objects with different electrical charges coming into contact with each other. An electrical discharge is created to neutralize this difference, which can damage electronic components and printed circuit boards. The following measures are generally sufficient to neutralize this difference before contact is made to protect your equipment from ESD.

- Use a grounded wrist strap designed to prevent static discharge.
- Discharge static electricity from your body by touching a known grounded surface, such as a computer's metal chassis.
- Keep all components and printed circuit boards (PCBs) in their antistatic bags until ready for use.
- For grounding purposes, make sure your computer chassis provides excellent conductivity between the power supply, the case, the mounting fasteners and the serverboard.

OPERATING PRECAUTIONS

$\begin{bmatrix} -\frac{3}{2} \end{bmatrix}$ Caution Risk of Explosion when battery is replaced by an incorrect type.

When installing the serverboard (CMOS) battery, there is a danger of explosion if the onboard battery is installed upside down, which will reverse its polarities (see figure). The battery must be replaced only with the same or an equivalent type recommended by the manufacturer. Dispose of used batteries according to the manufacturer's instructions.



INDEX

A

accessory kit contents 11 active-backup/failover network mode 52 adaptive load balancing network mode 52 administrator group adding to 61 editing 63 permissions 61 removing users from 62 administrator password, recovering 64 agency compliance 107 to 109 agency declarations Australia and New Zealand 109 Canadian Verification 108 China RoHS 109 European Community 107 FCC notice 108 GS mark 107 Japanese Verification 108 Underwriters Laboratory 107 WEEE 109 authentication best practice for setting security 33 CHAP 33 hosts 33 auto-insert option 57 80 autoloader option

B

back panel components 5 battery warning 114

С

CHAP 33 enabling in the Remote Management Console 35 Microsoft iSCSI initiator 37 Open-iSCSI for Linux 40 secret/password requirements 35 Xtend SAN initiator for Mac 42 cloning data 87 configuration files exporting 98 importing 98 configuring network settings 50 connecting ethernet and power cables 17 connecting to the Remote Management Console 20 converting logical device type 48

D

date and time setting 49 diagnostic error codes 103 LCD panel test sequence 94 performing an LCD panel test 93 RDX dock test 95 system logs 96 disk autoloader option 80

E

eject button/power LED indicators 7 ejecting RDX cartridges 86 electrical safety precautions 113 electrostatic discharge (ESD) precautions 113 email notification 53 customizing 54 environmental specifications 106 erasing RDX cartridges 86 error codes 103 establishing iSCSI connections 24 ethernet cables, connecting 17 exporting system configuration files 98

F

firmware automatic updates 54 updating manually 99 formatting RDX cartridges 86 front panel components 4

G

general safety precautions 112 groups adding users 61 editing user information 63 permissions 61 removing users from 62

Η

host authentication enabling in the Remote Management Console 36 overview 33

I

import/export tape options 57 importing system configuration files 98 installation configuring a static IP address via network.conf file 18 connecting network and power cables 17 iSCSI initiator software prerequisite 10 preparing for 10 rackmounting the RDX QuikStation 12 to 16 iSCSI best practice for setting security 33 configuration overview 24 connecting iSCSI targets for Linux 28 connecting iSCSI targets for Linux with CHAP 40 connecting iSCSI targets for Macintosh 30 connecting iSCSI targets for Macintosh with CHAP 42 connecting iSCSI targets for Windows 25 connecting iSCSI targets for Windows with CHAP - 37 enabling CHAP in the Remote Management Console 35 enabling host authentication in the Remote Management Console 36 managing connections 24 security overview 33 viewing active disk or virtual tape drive connections 75 viewing active library/autoloader connections 77 iSCSI initiator software prerequisite 10

L

LCD diagnostic test 93 LCD panels language option setting 54 type of information displayed 6 LED indicator panel functionality 7 Linux command to delete a target 29 command to disconnect a target 29 connecting iSCSI targets 28 connecting targets with CHAP 40 lithium battery warning 114 log files, generating 96 log level, changing 54 logical device overview 70 removable disk and virtual tape drive properties 74 virtual library/autoloader properties 76 logical device type configuring 21 converting 48 logically moving a tape cartridge 78 LTO-3 tape drive viewing active iSCSI connections 75 LTO-3 tape drives logical device properties 74 physical device properties 84 virtual configuration 70

Μ

Macintosh connecting iSCSI targets 30 connecting targets with CHAP 42 maintenance tasks 92 manager group adding to 61 editing 63 permissions 61 removing users from 62 Microsoft iSCSI initiator connecting with CHAP 37 connecting without CHAP 25

Ν

network configuring RDX QuikStation 50 port bonding options 52 network.conf file, configuring 18

0

Open-iSCSI for Linux command to delete a target 29 command to disconnect a target 29 connecting with CHAP 40 connecting without authentication 28 operating system precautions, battery warning 114

P

password.reset file, creating 66 passwords recovering for the built-in Admin account 64 requirements for CHAP 35 requirements for user groups 62 resetting Admin password via USB 66 permissions for groups in Remote Management Console 61 physical device properties 84 physical specifications 106 port bonding modes 52 power button, location 5 power ratings 106 preparing for installation 10 product features 2 product warranty caution iii properties removable disk and virtual tape drive 74 virtual library/autoloader 76

R

rackmount instructions 12 to 16 RDX cartridge cloning data 87 eject button/power LED indicators 7 ejecting 86 enabling unsafe eject 56 erasing/formatting 86 logically moving 78 RDX dock performing a diagnostic test 95 properties 85 RDX QuikStation accessory kit contents 11 configuring network information 50 connecting to the Remote Management Console 20 dock LCD panels 6 error codes 103 exporting configuration files 98 features 2 generating system logs 96 importing configuration files 98 LCD diagnostic test 93 LED indicator panel 7

RDX QuikStation, continued logical device type overview 70 physical device properties 84 physical specifications 106 power button location 5 power ratings 106 RDX dock diagnostic test 95 recommended system settings 22 restarting 97 safety guidelines 112 to 114 shutting down 97 system components 4 system maintenance overview 92 system settings 48 updating firmware 99 regulatory agency compliance 107 to 109 **Remote Management Console** adding users to a group 61 configuring logical device type 21 configuring network information 50 connecting to 20 editing user information 63 group permissions 61 iSCSI management rules 24 iSCSI security overview 33 logical device type overview 70 physical device properties 84 removing users 62 setting iSCSI security 34 system maintenance overview 92 system settings overview 48 viewing active disk or virtual tape drive iSCSI connections 75 viewing active iSCSI library/autoloader connections 77 removable disk configuration 70 logical device properties 74 physical device properties 84 unsafe eject mode 56 viewing active iSCSI connections 75 restarting the RDX QuikStation 97

S

safety agency compliance 107 to 109 safety guidelines battery warning 114 electrical precautions 113 ESD precautions 113

security enabling CHAP in the Remote Management Console 35 specifying hosts in the Remote Management Console 36 service, contacting Tandberg Data iii shutting down the RDX QuikStation 97 specifications agency compliance 107 to 109 physical 106 power 106 power ratings 106 SSH (Secure Shell) remote access option 54 system components 4 system maintenance exporting configuration files 98 generating system logs 96 importing configuration files 98 LCD panel diagnostic test 93 LCD panel test sequence 94 overview 92 RDX dock diagnostic test 95 restarting the RDX QuikStation 97 shutting down the RDX QuikStation 97 updating firmware 99 system settings 48 adding users to a group 61 automatic firmware updates 54 date/time 49 editing user groups 63 email notification 53 enable SSH remote access 54 group permissions 61 LCD panel language option 54 log level 54 network configuration 50 port bonding options 52 recommended 22 removing users 62 unsafe eject option 56

T

Tandberg Data LTO StorageLoader auto-insert option 57 import/export tape option 57 logically moving a tape cartridge 78 physical device properties 84 unique inquiry option 80 viewing active iSCSI connections 77 virtual configuration 70 virtual device properties 76 Tandberg Data T24 library auto-insert option 57 import/export tape option 57 logically moving a tape cartridge 78 physical device properties 84 unique inquiry option 80 viewing active iSCSI connections 77 virtual configuration 70 virtual device properties 76 Tandberg Data T40 library auto-insert option 57 import/export tape option 57 logically moving a tape cartridge 78 physical device properties 84 unique inquiry option 80 viewing active iSCSI connections 77 virtual configuration 70 virtual device properties 76 Tandberg Data, contacting iii tape autoloader option 80 technical support, contacting iii time setting, modifying 49 troubleshooting error codes 103 LCD diagnostic test 93 94 LCD panel test sequence RDX dock diagnostic test 95 system logs 96

U

unique inquiry option 80 unsafe eject option 56 updating firmware 99 USB flash drive configuring network information 18 generating system logs 96 resetting the administrator password 66 updating firmware 99 uses for 101 user group adding to 61 editing 63 permissions 61 removing users from 62

V

virtual autoloader option 80 virtual library auto-insert option 57 import/export options 57 logically moving a tape cartridge 78 physical device properties 84 unique inquiry option 80, 80 viewing active iSCSI connections 77 virtual library/autoloader logical device properties 76

W

Windows connecting iSCSI targets 25 connecting targets with CHAP 37

X

Xtend SAN iSCSI initiator for MAC connecting with CHAP 42 connecting without authentication 30

Notes